

NOTA DI LAVORO 139.2010

Does the Cause of Death Matter? The Effect of Dread, Controllability, Exposure and Latency on the Vsl

By **Anna Alberini**, University of Maryland, College Park, FEEM, Venice, Italy and Queen's University Belfast, Northern Ireland **Milan Ščasný**, Charles University Prague

SUSTAINABLE DEVELOPMENT Series Editor: Carlo Carraro

Does the Cause of Death Matter? The Effect of Dread, Controllability, Exposure and Latency on the Vsl

By Anna Alberini, University of Maryland, College Park, FEEM, Venice, Italy and Queen's University Belfast, Northern Ireland Milan Ščasný, Charles University Prague

Summary

The Value of a Statistical Life is a key input into the calculation of the benefits of environmental policies that save lives. To date, the VSL used in environmental policy analyses has not been adjusted for age or the cause of death. Air pollution regulations, however, are linked to reductions in the risk of dying for cancer, heart disease, and respiratory illnesses, raising the question whether a single VSL should be applied for all of these causes of death. We conducted a conjoint choice experiment survey in Milan, Italy, to investigate this question. We find that the VSL increases with dread, exposure, the respondents' assessments of the baseline risks, and experience with the specific risks being studied. The VSL is higher when the risk reduction is delivered by a public program, and increases with the effectiveness rating assigned by the respondent to public programs that address specific causes of death. The effectiveness of private risk-reducing behaviors is also positively associated with the VSL, but the effect is only half as large as that of public program effectiveness. The coefficients on dummies for the cause of death per se-namely, whether it's cancer, a road traffic accident or a respiratory illness-are strongly statistically significant. All else the same, the fact that the cause of the death is "cancer" results in a VSL that is almost one million euro *above* the amount predicted by dread, exposure, beliefs, etc. The VSL in the road safety context is about one million euro less than what is predicted by dread, exposure, beliefs, etc. These effects are large, but the majority of the variation in the VSL is accounted for by the public program feature, the effectiveness of public programs at reducing the indicated risk, and dread. The effects of exposure and experience are smaller. These results raise the question whether using VSL figures based on private risk reduction, which is usually recommended to avoid double-counting, severely understates how much a society might be willing to pay for public safety.

Keywords: VSL, Conjoint Choice Experiments, Mortality Risk Reductions, Cost-benefit Analysis, Forced Choice Questions

JEL Classification: 118, J17, K32, Q51

Address for correspondence:

Anna Alberini AREC, 2200 Symons Hall University of Maryland College Park, MD 20742 USA E-mail: aalberini@arec.umd.edu

DOES THE CAUSE OF DEATH MATTER? THE EFFECT OF DREAD, CONTROLLABILITY, EXPOSURE AND LATENCY ON THE VSL

By

Anna Alberini* University of Maryland, College Park FEEM, Venice, Italy and Queen's University Belfast, Northern Ireland

and

Milan Ščasný Charles University Prague

Last revision: 20 October 2010 Last revision by: Anna Alberini

*: Corresponding author. Please address correspondence to Alberini, AREC, 2200 Symons Hall, University of Maryland, College Park, MD 20742, e-mail: <u>aalberini@arec.umd.edu</u>.

This research was conducted as part of the VERHI-Children Research Program, funded by the European Commission, SSPE-CT-2005-6529, http://www.oecd.org/site/0,3407,en_21571361_36146795_1_1_1_1_1_00.html

Abstract

The Value of a Statistical Life is a key input into the calculation of the benefits of environmental policies that save lives. To date, the VSL used in environmental policy analyses has not been adjusted for age or the cause of death. Air pollution regulations, however, are linked to reductions in the risk of dying for cancer, heart disease, and respiratory illnesses, raising the question whether a single VSL should be applied for all of these causes of death. We conducted a conjoint choice experiment survey in Milan, Italy, to investigate this question.

We find that the VSL increases with dread, exposure, the respondents' assessments of the baseline risks, and experience with the specific risks being studied. The VSL is higher when the risk reduction is delivered by a public program, and increases with the effectiveness rating assigned by the respondent to public programs that address specific causes of death. The effectiveness of private risk-reducing behaviors is also positively associated with the VSL, but the effect is only half as large as that of public program effectiveness.

The coefficients on dummies for the cause of death per se—namely, whether it's cancer, a roadtraffic accident or a respiratory illness—are strongly statistically significant. All else the same, the fact that the cause of the death is "cancer" results in a VSL that is almost one million euro *above* the amount predicted by dread, exposure, beliefs, etc. The VSL in the road safety context about is about one million euro *less* than what is predicted by dread, exposure, beliefs, etc.

These effects are large, but the majority of the variation in the VSL is accounted for by the public program feature, the effectiveness of public programs at reducing the indicated risk, and dread. The effects of exposure and experience are smaller. These results raise the question whether using VSL figures based on private risk reduction, which is usually recommended to avoid double-counting, severely understates how much a society might be willing to pay for public safety.

- **JEL classification:** I18 (Government Policy; Regulation; Public Health); J17 (Value of Life; Forgone Income); K32 (Environmental, Health, and Safety Law); Q51 (Valuation of Environmental Effects)
- **Keywords:** VSL; conjoint choice experiments; mortality risk reductions; cost-benefit analysis; forced choice questions.

Does the Cause of Death Matter? The Effect of Dread, Controllability, Exposure and Latency on the VSL

by Anna Alberini and Milan Ščasný

1. Introduction.

There is a reasonable degree of consensus in academic and policy circles that the Value of a Statistical Life (VSL) is the appropriate metric to estimate the mortality benefits of policies that reduce premature mortality, such an environmental and safety programs. There is much less agreement as to whether a single VSL figure should be used for all beneficiaries and for all causes of death covered by the policy.

In US environmental policy assessments, for example, analysts typically rely on estimates of the VSL based on labor market studies (US EPA, 2000; Viscusi, 1993; Viscusi and Aldy, 2003, and Aldy and Viscusi, 2007). Questions have been raised whether such practice is appropriate, since the beneficiaries of environmental regulations are usually the very old (Krupnick, 2007) or the very young, and the causes and timing of death are very different from workplace accidents.

Consider now air pollution. Recent research (US EPA, 1999a, 1999b, Hurley et al., 2005, National Academy of Science, 2008) indicates that the most important mortality effects of air pollution are those associated with cardiovascular disease, followed by cancer. Air pollution is also thought to trigger asthma attacks in asthmatic subjects, exacerbate the severity of asthma attacks, and increase asthma- and chronic obstructive pulmonary disease-related mortality.

Should a single VSL be used for such diverse mortality effects? Economic theory suggests a number of reasons why individuals might place a different value on them. Previous psychometric research further indicates that individual perceive risks along many dimensions,

including voluntariness, controllability, and dread (Starr, 1969; Fischhoff et al., 1978, and Slovic, 1987), and such perceptions may influence their willingness to pay to reduce risks (McDaniels et al., 1992). In policy practice, the US Environmental Protection Agency uses a single VSL in its policy analyses.¹ By contrast, the Directorate-General Environment of the European Commission recommends a VSL figure that is 50% higher for cancer deaths than for other causes of death.²

In this paper, we report on the results of a choice experiment study that was specifically designed to investigate this issue. In our choice experiments, we created hypothetical alternatives defined by five attributes: i) the cause of death (respiratory illness, cancer, or road-traffic risks), ii) the size of the risk reduction, iii) whether the risk reduction was private or delivered by a public program, the latter case implying that there are other beneficiaries, iv) latency, expressed as the number of years until the risk reduction occurs, v) the one-time cost to the respondent, which must be paid now.

Respondents were to indicate their most preferred alternative out of a choice set that included two hypothetical alternatives and the status quo.³ We use the responses to the choice questions to estimate the VSL, i.e., the Willingness to Pay (WTP) for a unit risk reduction. The conjoint choice experiment survey was administered to a sample of residents of the city of Milan, Italy, in late November to mid-December 2008.

Regarding i), we chose respiratory illness and cancer because these are risks associated with air pollution (and, in the case of cancer, other environmental exposures, including

¹ In 2000, the Science Advisory Board-- Environmental Economics Advisory Committee to the US EPA advised the agency not to adjust the VSL (Dockins et al. 2004).

² See <u>http://ec.europa.eu/environment/enveco/others/pdf/recommended_interim_values.pdf</u>.

³ Half of the sample was further asked a forced-choice question, in which they were to choose between two hypothetical alternative without being given an opt-out response category.

contaminated water and soil). We also focus on road traffic risks because virtually everyone is familiar with them, they can be addressed through both private behaviors and public programs, and people may hold various degrees of confidence about the controllability of such risks. Since i) and iv) are varied independently of one another, our study design allows us to investigate whether the cause of death has an effect on the willingness to pay to reduce risk that is above and beyond the mere futurity of the risk reduction.

We use attribute iii)—the private or public program nature of the risk reduction combined with the respondent's assessment of the effectiveness of private behaviors and public program in reducing each of the three types of risk here studied, to examine whether the controllability of a risk influences the WTP to reduce such a risk. We elicit dread directly from the respondents.

McDaniels et al. (1992) introduce personal exposure as a determinant of the VSL, and find that the WTP to reduce a particular mortality risk increases when individuals feel personally exposed to it. We control for baseline risk as stated to the respondent in the survey, respondent assessment of baseline risks, personal exposure, and experience with the risk.

Briefly, we find that the VSL does increase with dread, with our construct of the respondent's exposure to the three types of risk, and with the respondents' assessments of the baseline risks and experience with the risks. The VSL is higher when the risk reduction is delivered by a public program, and increases with the effectiveness rating assigned by the respondent to public program in addressing that cause of death. The effectiveness of private risk-reducing behaviors is also positively associated with the VSL, but the effect is only half as large as that of public program effectiveness.

In our models, we control for the characteristics of the mortality risk reductions respondents were to value. Yet the cause of death per se—namely, whether it's cancer, a road-traffic accident or a respiratory illness—remains strongly statistically significant. All else the same, the fact that the cause of the death is "cancer" results in a VSL that is almost one million euro *above* what is predicted by dread, exposure, beliefs, etc. The VSL in the road safety context about is just over one million euro *less* than what is predicted by dread, exposure, beliefs, etc.

These effects are large, but the majority of the variation in the VSL is accounted for by the public program feature, the effectiveness of public programs in reducing the indicated risk, and dread. To illustrate, the VSL in a public program context is $\in 2$ million more than when the risk reduction is private, changing the perceived effectiveness of the government program from the lowest to the highest value increases the VSL by just under $\in 2$ million, and a change from the lowest to the highest dread raises by the VSL by $\in 2$ million. The effects of exposure, experience, etc. are smaller. These results raise the question whether using VSL figures based on private risk reduction, which is usually recommended to avoid double-counting, may severely understate how much a society might be willing to pay for public safety (see Nielsen et al., 2009).

The remainder of this paper is organized as follows. Section 2 presents background information and reviews the literature. Section 3 describes the research questions and study design. Section 4 presents the questionnaire and survey administration. Section 5 presents the theoretical and econometric model. Section 6 discusses the data. Section 7 presents the estimation results, and section 8 concludes.

2. Background and Previous Literature.

A. The VSL and the Cause of Death

The Value of a Statistical Life is defined as the marginal WTP for a small change in the risk of dying:

(1)
$$VSL = \frac{\partial WTP}{\partial R}\Big|_{U=const.}$$

As a summary measure of the WTP for mortality risk reductions, the VSL is used to compute the monetized benefits of policies that save lives.⁴ Implicit in (1) and in most standard expected utility models is the notion that R represents the total risk of dying for any cause.

Economic theory suggests several reasons why the VSL for one cause of death might be different from that for another. For starters, the VSL should increase with baseline risks (Pratt and Zeckhauser, 1996). All else the same, the VSL for a specific cause of death might be larger simply because the baseline risk of dying for that cause is higher.

The existence of competing risks might be another reason for different VSLs. Eeckhoudt and Hammitt (2001) consider competing risks and show that if the utility of a bequest at death is positive, then the marginal WTP for reducing one type of risk (i.e., the VSL for that cause of death) depends on the magnitude of the other risks of dying. Based on their model, a person in poor health with a high risk of dying from a chronic illness would have a very low willingness to pay for a small reduction in the risk of dying in a car accident or because of pollution exposures, which account for very small shares of this person's total risk of dying. Since a large competing risk reduces the chance to benefit from a reduction in the specific risk, Eeckhoudt and Hammitt dub this the "why bother" effect. Evans and Smith (2006) show that the effect of a competing risk is potentially ambiguous, because it depends on how the competing risks enter in the expected utility.

⁴ By "saving lives," we mean "reducing or eliminating premature deaths."

Another reason why people might be willing to pay different amounts of money to reduce the risk of dying from different causes may simply lie in the timing of the risk reduction. Economic theory shows that the VSL at time t for a risk reduction to be incurred L periods later is equal to the VSL for an immediate risk reduction in period (t+L), discounted back to the present (Cropper and Sussman, 1990).

B. Perceptions

There are two main questions related to risk perceptions: a) Why are certain hazards considered worse than others? and b) Do some people consider certain hazards worse than others? The answer to the first question comes from the psychometrics literature, which shows that risk perceptions are influenced by the attributes of the risk beyond its sheer magnitude (e.g., its controllability, familiarity, dread, and whether it is voluntarily faced or not) (Fischhoff et al., 1978; Slovic, 1987; Chauvin et al., 2007, and Urban and Ščasný, 2007).

It is possible that such differences in perceptions influence the WTP to reduce the various types of mortality risks, even holding the magnitude of the risks and latency the same (Revesz, 1999; Rowlatt et al., 1998). For example, evidence from surveys suggests that people consider it very important to reduce cancer deaths (e.g., Jones-Lee et al., 1985), and might be willing to commit more resources to reduce risks with which they are not familiar and/or they consider outside of their own control (McDaniels et al., 1992, Savage, 1993a, and Rowlatt et al., 1998). McDaniels et al. (1992) introduce personal exposure as a determinant of WTP to reduce risks, and find that the WTP increases if a respondent is personally exposed to a particular risk of death.

The answer to the second question is less straightforward, in part because capturing individual characteristics that explain risk perception has turned out to be very difficult. Urban and Ščasný (2007) review empirical studies and group possible determinants of risk perceptions into several categories: socio-demographic characteristics, religious and quasi-religious beliefs, general trust level, cultural factors, personal facets, experience or information learning process.

Debate continues on the magnitude of the effects attributable to perceptions. While perceptions did affect respondents' priorities over safety programs, Chilton et al. (2002) conclude that "the impact of these perceptions is a great deal less pronounced than has been by the value differentials that are currently implicit—and in some cases, explicit—in public policy making."

C. Survey Effects and Aspects of the Provision of the Risk Reduction

In stated preference studies, the VSL is approximated from the WTP for a finite reduction in mortality risks or is inferred from the tradeoffs between hypothetical risk reduction profiles made by survey respondents (Alberini, 2005). Aspects of the scenario may influence the WTP, and hence the VSL.

Stated preference studies about mortality risk reductions need to devise a credible mechanism for delivering the risk reduction. In many cases, the most plausible or appropriate mechanism for delivering risk reductions is a public program. One problem with this approach, however, is that the respondents' altruistic considerations may result in double-counting. Economic theory (Jones-Lee, 1991, 1992) has worked out the conditions under which double-counting will and will not occur, showing that they depend crucially on (i) the type of altruism

affecting the responses (paternalistic or non-paternalistic), and (ii) what the respondent is told to assume about the payments made by other people.

Unfortunately, in applied work it is very difficult to observe the nature of each respondent's altruism, and efforts to tell respondents what to assume about other people's payments have proven awkward and confusing (Johannesson and Meltzer, 1998). This has prompted many researchers to turn to valuing private risk reductions (e.g., Dickie and Gerking, 1996, Krupnick et al., 2002), even though this is likely to produce only a lower bound for WTP.⁵

Altruistic considerations are not the only reason for differences in WTP across public and private risk reductions: Respondents may also attach a different probability of provision and/or effectiveness of the risk reduction to government programs and private actions. Valuation through stated preference studies is also affected by experience with the good to be valued and with the valuation task itself (Bateman et al., 2008).

D. Empirical Evidence

Since environmental exposures are often linked with cancer health endpoints, much research has focused on whether the cancer VSL is different than that for other causes of death. Cancer is associated with suffering and pain, and is highly dreaded (see Starr, 1969, Fischhoff et al., 1978; Slovic, 1987), which is often taken to imply that the VSL should be greater when the cause of death is cancer (Revesz, 1999; US EPA 2000). Some studies have found that people favor programs that reduce cancer mortality (e.g., Jones-Lee et al., 1985; Mendeloff and Kaplan,

⁵ Private risk reductions are generally thought to result in conservative estimates of the VSL, but we are aware of at least one study (Johannesson et al., 1996b) that actually found them to be *larger* than the VSL estimate for a comparable risk reduction in a public program context.

1989; McDaniels et al., 1992; Savage, 1993, and Tolley et al., 1994), and others report no such cancer "premium."

In Magat et al. (1996), the median respondent was indifferent between reducing the risk of terminal lymph cancer and reducing the risk of automobile death, whereas terminal lymph cancer risk is about 1.33 worse than the risk of curable lymph cancer (and automobile death). Using a similar approach (risk-risk questions), Van Houtven et al. (2008) find that individuals have a strong preference for avoiding cancer risks, but that this preference wanes as the cancer latency period increases.

None of the abovementioned studies estimated the VSL directly. The few studies that have attempted to value mortality risk reductions have found surprisingly little evidence that the cancer VSL is higher than the VSL for other causes of death. In a contingent valuation study in Taiwan, Hammitt and Liu (2004) find that the WTP to reduce the risk of cancer is about one-third larger than that to reduce the risk of a similar chronic, degenerative disease. However, the coefficient on the cancer dummy is not significant at the conventional levels. Likewise, Hammitt and Haninger (2010) elicit the WTP to reduce fatal-disease risks in adults and children caused by consuming pesticide residue in foods. They find that the WTP for cancer and non-cancer diseases is similar to the WTP to reduce motor-vehicle crashes.

One possible reason for these empirical findings is that low baseline risks may offset the effect of dread (or other perceptions) on the VSL. Chilton et al. (2006) use a variant on the "risk-risk" approach to identify the separate effect of (contextless) baseline risk and dread effects for various risks, including that of dying in a pedestrian accident, in an automobile driver/passenger accident, or by murder. The respondents in their study reported substantial dread for certain types

of death (especially for rail accident, fire in a public place, and drowning), which in some cases was cancelled out by the low baseline risks.

Tsuge et al. (2005) conducted conjoint choice experiments where the alternatives are defined by four attributes: cost, the size of the risk reduction, the type of risk (all causes of death, accident, cancer and heart disease) and latency. Tsuge et al. elicit the respondent's perceptions along many dimensions, including what they term "controllability" ("Government can reduce this risk"), "dread" ("Pain accompanies this risk" and "This risk is terrible."), a subjective assessment of exposure, and subjective assessments of public and private knowledge. Tsuge et al. conclude that "it is almost unnecessary to adjust the VSL according to the difference in the type of risks if the VSL is adequately calculated."

3. Research Questions and Study Design.

A. Research Questions and Approach

In this paper, we focus on three research questions. First, all else the same, does the VSL vary with the cause of death? Second, all else the same, are people willing to pay different amounts of money when the risk reductions are delivered by public programs? Third, are the differences in the VSL by cause of death (if any) explained away by latency, dread, controllability, exposure and salience of the risks to the respondents? And how large is the contribution to the VSL from each of these factors?

To answer these questions, it is important to use a single valuation method (e.g., conjoint choice experiments) to avoid confounding true differences in VSL with valuation methodinduced effects. We chose conjoint choice experiments (see Bateman et al., 2002) because by varying risk attributes across and within respondents, we can disentangle the effect of such variations on the responses. We developed a questionnaire based on conjoint choice experiments that included several "treatments," and administered it to a sample of residents of Milan, Italy. We conducted the study at one locale to reduce heterogeneity due to unobservables.

Using the responses to the conjoint choice questions, we examine whether the VSL is systematically related to risk attributes—those assigned by design to the respondent as well as the respondent's subjective assessments—and individual characteristics of the respondents. We also attempt to quantify the portion of the VSL that is accounted for by each of these factors.

B. Estimating the VSL Using Conjoint Choice Experiments

Choice experiments are a stated preference method, and as such they rely on what people *say* they would do under hypothetical, but realistic, circumstances to place a value on non-market goods (see Bateman et al., 2002). An advantage of stated preference approaches over revealed preference approaches is that the former do not assume that people's quantitative and qualitative subjective risk perceptions are the same as the objective risks. Respondents can be educated about existing risk levels and about opportunities for mitigating risks.

Stated preference approaches can cater to a variety of causes of death, beneficiaries of the risk reduction, and latency. Another important advantage is that they allow analysts to circumvent the lack of variation in risk reductions that can plague revealed preference studies, as well as the correlation between cause of death, latency and other risk attributes, which makes it difficult to interpret results. Disadvantages include the possibility of hypothetical bias, and cognitive difficulties affecting respondents' processing of small probabilities.

In conjoint choice experiments, respondents are asked to select one of K hypothetical scenarios ($K \ge 2$), where the scenarios are described by a vector of attributes, including cost and,

in this study, mortality risk reductions. Individuals are assumed to pick the alternative to which they attach the highest utility. Conjoint choice experiments were used to value mortality or cancer risk reductions in Tsuge et al. (2005), Itaoka et al. (2006), Alberini et al. (2007), and Tonin et al. (2009).⁶

C. Our Conjoint Choice Experiments

We started the conjoint choice experiment portion of our questionnaire by informing the respondent that he or she would be the beneficiary of the risk reductions we were about to describe (and the sole beneficiary, in the case of private risk reductions).

The hypothetical alternatives in the conjoint choice experiments were described by 5 attributes: (i) the cause of death (respiratory illnesses, cancer, road traffic accidents), (ii) whether the risk reduction is attained by a public program or is private, (iii) the risk reduction itself, (iv) latency, expressed as the number of years that must elapse before the risk reduction begins, and (v) cost. Attributes and attribute levels are summarized in table 1. A sample conjoint choice question is reproduced in Figure 1.

Regarding (i), we focused on cancer and respiratory causes of death because these have been linked to environmental exposures and are addressed by many environmental programs. We also include mortality risks for road traffic accidents for three reasons. First, virtually everyone uses the roads, and people are generally aware of road traffic risks, so scenarios focusing on such

⁶ Contingent Valuation (CV) is another stated preference method that is sometimes used to estimate the VSL. In CV surveys about risk reductions, respondents are usually queried directly about their willingness to pay for a public program, product or good that reduces their risk of dying. CV surveys were used, among others, by Johannesson et al. (1996a), Johannesson et al. (1997), Krupnick et al. (2002), and Alberini and Chiabai (2007a, 2007b) for samples of adults and the elderly. Dickie and Gerking (2006) present the results of a CV survey that elicits WTP to reduce the risk of developing fatal skin cancer in adults and children.

risks are salient and plausible to most people. Second, they can be addressed through both individual behaviors and public programs.

Third, there is no question that most people would regard road traffic risks as familiar and controllable (at least to some extent). They are thus well suited for the purposes of our survey, and can serve as a useful comparison with less common risks (such as respiratory risks) and risks that are accompanied by morbidity, pain and dread (cancer risks). In earlier stated-preference studies conducted in other countries, such as the U.K. (Jones-Lee, 1989), Sweden (Johannesson et al., 1996b; Persson et al., 2001) and India (Bhattacharya et al., 2007), people were willing and capable to trade off money for road traffic risk reductions.

Public v. private risk reductions (item (ii) above) were presented to the respondents with a reminder that the former imply that there are other beneficiaries of the risk reduction beyond the respondent, whereas the respondent is the sole beneficiary of the risk reduction when the action is private. We described the public programs as being "nationwide."

Our interest in the public v. private nature of risk reduction is driven by two reasons. First, much earlier stated-preference research has asked people to value private risk reduction in hopes of avoiding double-counting (see section 2), but environmental and other safety regulations are part of public programs, and it is of interest to determine how widely the estimates of the VSL based on private risk reductions differ from those with a public program. Second, private behaviors v. public programs provide us with an opportunity to study if controllability of risks—which depends on how effective people judge the risk-reducing measures—influences the WTP to reduce them.

Latency is expressed as the number of years that elapse before the risk reduction begins. To avoid confounding between the cause of death and the latency aspect, we used latency levels of 0 (=immediate risk reduction), 2, 5 and 10 years, and we varied this attribute independently of the context of death and the other attributes.

We used four possible levels for the risk reduction, namely 2, 3, 5 and 7 in 10,000 over 5 years. Finally, each alternative risk reduction plan had a price tag. This cost would be incurred by the respondent's household immediately and would be paid this one time only. We used four possible cost amounts ranging from 200 to 2000 (see table 1). Under alternative assumptions about the discount rates, these cost amounts correspond to VSL of a few hundred thousand to several million euro.

Each respondent was randomly assigned to a set of 5 pairs of risk reduction profiles. There were a total of 32 possible sets, and we imposed certain restrictions on them to ease the respondent's task. For example, the first two pairs viewed by the respondents (profiles A and B, and C and D) focused on the same cause of death, which was selected at random between the three studied in this project. In addition, within each pair, the latency period was restricted to be the same for both alternatives to keep the respondent's task manageable. Identification of the discount rate relies on within- and between-respondent variation in the time horizon when the risk reduction would be realized.^{7, 8}

⁷ These 32 sets of pairs were selected at random and without replacements from the full universe of non-dominated pairs that satisfied all of the abovementioned requirements. Due to a software error, however, the last pair in set 8 contained a dominated choice. In the analyses reported in this paper, for good measure we check the robustness of results of the non-linear conditional logit after deleting the responses to the questions about this pair from the sample.

⁸ We also created an additional treatment whereby about one half of the respondents first faced a forced choice question (choose between A and B), and then were asked which they would prefer between A, B and the status quo (no payment and no risk reduction). The remainder of the respondents was asked to choose directly between A, B and the status quo. Assignment to one or the other variant of this treatment is random. Respondents assigned to treatment TFORMAT=1 thus engaged in a total of $2\times5=10$ conjoint choice tasks each; respondents assigned to TFORMAT=2 engaged in $1\times5=5$ conjoint choice tasks each. The purpose of this split sample treatment is to check whether the forced choice exercise induced response effects that changed the VSL. In practice, we found that it did not (see Alberini et al., 2009), and for this reason in this paper we pool the responses from both versions of the questionnaire.

4. Questionnaire, Sampling Frame and Survey Administration.

A. Structure of the Questionnaire

The questionnaire is self-administered by the respondent using the computer. The computer interview starts with the respondent entering his or her gender, age, and the name, age, and gender of each of his or her children.

Section A asks questions about the health status of one of the respondent's children (selected at random among those aged 17 or younger), and section B about the respondent's own health. Section C of the questionnaire elicits extensive information about use of roadways, lifestyle, environment, genetic predisposition to cancer and familiarity with it. The purpose of this section is to understand the salience of certain risks to the respondent and to get a sense of the exposure to certain risk factors.

Section D contains a probability tutorial. We start with a simple and intuitive presentation based on tossing a coin or casting a die, but point out that the notion of chance also applies in other familiar situations (e.g., the weather forecast and the chance of rain). This is followed by a simple quiz to make sure that people have grasped the basics of probability.

We then move on to the notion of mortality risks. We use two visual representations of risk: (i) a grid with 10,000 squares, which we use when attention is restricted to a reference group or population,⁹ and (ii) bar charts, which we use when we want to show how risks vary across age groups (and hence change as a person ages).

⁹ When we first introduce the notion of risk of dying, we illustrate it using the grid of 10,000 squares, where white squares denote the people who survive and blue squares denote the people who die. We first display a handful of blue squares scattered at random on the grid to convey the notion of randomness, then the blue squares move on the grid until they are aligned in the northwest corner of the grid. The purpose of this animation is to convey the sense of the magnitude of the risks. Corso et al. (2001) examine the effect of differential visual aids on the sensitivity of willingness to pay to reduce risks.

In section E of the questionnaire we inform the respondent that it is possible to reduce one's own risk of dying in many ways. Using respiratory illnesses, cancer and road traffic accidents as examples, we explain that risk reductions may result from individual actions (e.g., getting a flu shot, purchasing a car with safety equipment) and government programs (e.g., an air pollution control program). We also emphasize that some actions are specific for men (e.g., prostate cancer tests), some are specific for women (e.g., pap smears), and others apply only to children (child seats in cars).

Section F contains an exercise that strips risks of all other attributes and makes respondent focus on the magnitude of the risks. In section G of the questionnaire we zero in on the three causes of death that are at the heart of this questionnaire, namely cancer, respiratory illnesses, and road-traffic accidents. In addition to providing some basic information about them, we also ask for people's subjective assessments of the comparative magnitude of these risks for people their age. In section H, the respondents express their opinions on the effectiveness of private actions and public programs in reducing the risk of dying for each of the three causes of death studied in this project, and also assess the futurity v. immediateness of risks.

Section I is dedicated to the conjoint choice questions. In the first screen of this section, we summarize the five attributes of the alternatives being compared, remind respondents of their own baseline risks, and point out that the choice experiments will contain relatively small risk reductions.

After the conjoint choice questions, we ask debriefing questions and explore reasons for the observed choices. The final section of the questionnaire elicits information about the respondent's socio-economic circumstances.

B. Administration of the Survey and Sampling Plan

As mentioned, our questionnaire is self-administered by the respondent using the computer. We chose this option, instead of in-person interviews, for cost reasons and because our study design involves numerous treatments, questionnaire variants and visuals. Respondents took the survey in two facilities in Milan, Italy, and were paid $\in 10$ for their participation in the survey.

The final survey was preceded by two pilot studies, which were conducted in Milan in June (N=200) and September 2008 (N=100), respectively. Both pilots used the same sampling frame as the final survey. Our universe was Milan residents aged 20-60 who had at least one child of age 17 or younger.¹⁰ The sample was to be evenly divided among three age groups, namely persons aged 20-34, 35-44, and 45-60.

The sample was to have an even number of mothers and fathers, and to mirror the city's population in terms of education and income. For example, in Milan, 23% of the residents aged 20-60 have a college degree, and mean (after tax) household income is about €30,000 a year. We also specified that no more than 20% of the persons in the sample should be homemakers.

We chose to restrict attention to Milan, rather than to a nationally representative sample, for two reasons. For starters, limiting the survey to a single city was significantly less expensive and allowed us to increase the sample size, albeit at the expense of national representativeness. Second, Milan suffers from a serious air pollution problem. Residents are well aware of this and well informed about bad air pollution episodes, which are generally covered by the local news. Focus groups held in 2006 also suggested that Milan residents are well informed about the health

¹⁰ Our universe was parents with at least one legally minor child because this survey was part of a broader research project on parents' valuation of own or their children's mortality risk reductions. In this paper, attention is restricted to those adult respondents who had to engage in tradeoffs about their own risk of dying.

effects of air pollution and other types of environmental exposures (e.g., contaminated sites).¹¹ These considerations suggested that examples of public programs that reduce air pollution and hence mortality risks (mentioned in section G of the questionnaire) would be understood and accepted by the respondents.

5. The Model.

A. The Random Utility Model

We assume that the responses to the conjoint choice experiment questions are driven by a random utility model. We posit that the deterministic portion of the indirect utility function is:

(2)
$$\overline{V}_{ij} = \alpha \cdot DR_{ij} \cdot \pi(L) + \beta \cdot (y_i - C_{ij}),$$

where DR is the discounted risk reduction (see below), $\pi(L)$ is the probability of surviving L years, until the risk reduction begins, α is the marginal utility of a unit of risk reduction, β is the marginal utility of income, (y-C) is residual income, and subscripts i and j denote the individual and the alternative, respectively.

Assuming constant exponential discounting, the discounted risk reduction is defined as:

$$DR = \Delta R \cdot e^{-\delta \cdot L},$$

where ΔR is the risk reduction, L is the number of years that elapse before the risk reduction begins and δ is the discount rate.

On appending an error term, which captures aspects of the indirect utility that are known to the respondent but not the analyst, we obtain the random utility model:

¹¹ The evidence from the focus groups was confirmed by the results of a pen-and-paper questionnaire that participants in Pilot 1 took once they had completed their computer questionnaire.

(4)
$$V_{ij} = \overline{V}_{ij} + \varepsilon_{ij}.$$

In each conjoint choice experiment question, the respondent is asked to examine K alternatives and to indicate the most preferred option.¹² We assume that the respondent will choose the one with the highest indirect utility. If we further posit that the error terms in (4) are i.i.d. and follow a standard type I extreme value distribution, the probability that the respondent chooses alternative k is:

(5)
$$\Pr(k) = \frac{\exp(\overline{V_k})}{\sum_{j=1}^{K} \exp(\overline{V_j})}.$$

Expression (5), where we have omitted the subscript i to avoid notational clutter, is the contribution to the likelihood of a conditional logit model. The only difference with respect to the conventional conditional logit is that here \overline{V} is non-linear in the attributes and the parameters.

Since the VSL is the marginal utility of the risk reduction divided by the marginal utility of income, we estimate it as

(6)
$$VSL = \frac{\hat{\alpha}}{\hat{\beta}} \times 10,000,$$

where the hats denote maximum likelihood estimates. Multiplication by 10,000 is necessary because in our estimation routine we express the risk reduction as, say, 3 or 4 (in 10,000) instead of 0.0003 or 0.0004.

Equations (2)-(6) assume that the VSL is constant for all individuals in the sample, and that the cause of death or the source of the risk reduction does not matter. The model is easily

 $^{^{12}}$ K is equal to 2 in the forced choice questions, and to 3 in all other questions, which offer the status quo as well as two hypothetical risk reduction profiles.

amended to allow for the cause of death, and for the mechanism in which the risk reduction is attained (a public program versus a private behavior) to affect utility and to result in potentially different VSLs:

(7)
$$\overline{V}_{ij} = \left[DR_{ij} \cdot \pi(L) \right] \cdot \left[\alpha_1 + \alpha_2 \cdot CANC_{ij} + \alpha_3 \cdot RESP_{ij} + \alpha_4 \cdot PUBL_{ij} \right] + \beta \cdot (y_i - C_{ij})$$

where CANC, RESP, and PUBL are dummies denoting cancer, respiratory illnesses, and a public program, respectively, and the α s are marginal utilities.

Finally, we wish to examine whether subjective and researcher-assessed attributes and perceptions of risk influence the VSL. For this purpose, we enter in the second brackets of equation (7) additional variables capturing perceptions and individual characteristics of the respondents. We discuss them in the next subsection.

B. Additional Effects

We wish to investigate whether the VSL varies with the cause of death, and, if so, whether this effect is explained away by other attributes of the risk (including researcher- and respondent-assessed exposure, sensitivity, experience, and dread) and aspects of the provision of the risk reduction. To find out, we include in the model variables that measure (i) the effectiveness of public programs and private behaviors in reducing the stated risks, (ii) baseline risks, (iii) exposure, (iv) beliefs about baseline risks of dying for a specific cause of death for a person the respondent's age, (v) sensitivity, which depends on current health status, (vi) previous experience with the risks being valued, and (vii) dread.

To measure (i), we create dummies for whether the respondent indicated that he thinks that public programs are "very effective" at reducing the risk of dying for each of the three causes consider in the questionnaire. PUBEFF for alternative A will then take a value of one if the alternative posits that the risk reduction would apply to cause *h* and would be incurred through a public program, and the respondent rated public programs as "very effective" in reducing the risk of dying from cause *h*. PRIVEFF is similarly constructed for private behavior alternatives. These variables capture both aspects of the scenario and controllability of the risks.

To control for (total) baseline risk, we enter the age-specific risk of dying for all causes as we stated it to the respondent in the survey. EXPOSURE varies with alternative and the respondent. Based on the responses to section C of the questionnaire, if the respondent lives in a high pollution area and smokes, and if alternative A is about respiratory illness, then exposure for alternative A is equal to 1. We constructed similar measures if an alternative is about road traffic risks and the respondent reports that he or she uses the road every day and does not necessarily use seatbelts all the time. Exposure to cancer takes on a value of 1 when the respondent believes that cancer runs in his or her family, a blood-related immediate family member has or has cancer, and the alternative is about cancer risks.

Likewise, MORECOMMON varies with the alternative. In section G of the questionnaire, respondents were asked which were more common among people their age, deaths in road traffic accidents or for cancer. Suppose that a respondent indicates that he believes that deaths in road traffic accidents are more common among people his age than cancer deaths. If alternative A is about road traffic accidents and B about cancer, then MORECOMMON will be equal to one for alternative A and to zero for alternative B. (This variable is coded to zero for respiratory risks.)

To create SENSITIVITY, we consider a respondent "sensitive" to cancer if he does have cancer. We consider a respondent "sensitive" to respiratory illnesses if he reports having asthma, emphysema, chronic bronchitis or other chronic respiratory conditions, and these conditions

impair daily activities. We coded this variable to zero for road-traffic risks. We interpret this variable as the respondent's adjustment to the population risks indicated in MORECOMMON.

EXPERIENCE is coded as a one if the respondent has a relative, a spouse or close friend who has cancer (and the alternative in the conjoint choice task is about cancer). We also code it as a one when the respondent has had to go to the emergency room or be hospitalized as a result of a road traffic accident or a respiratory illness (and the alternative focus on road traffic risks or respiratory illnesses).

Regarding dread, recall that we asked respondents to rate the level of dread associated with several causes of death on a scale from 1 to 5, where 1 means no or minimal dread and 5 means the highest degree of dread. Suppose that a respondent rates the dread associated with road-traffic accident deaths as a 2 and that of cancer as a 4. If alternative A involves reducing risks in the road traffic accident context and B is a cancer risk reduction, then the DREAD assigned to each alternative will be 2 and 4, respectively.

We use about these alternative- and beneficiary-specific variables to test two main hypotheses. First, we wish to see if they enter in the model significantly and if their coefficients have the expected positive signs. If these expectations are borne out in the data, they suggest that the responses to the conjoint choice experiments are internally valid.

Second, *if* these variables capture all dimensions of risk, further labeling the risks as "risk of dying in road traffic accidents," "for cancer" and "for respiratory causes" should not convey any additional information, and dummies for the cause of death for the alternatives being considered would not be important predictors of the choice between the alternatives. This is an important issue for benefit transfer purposes, i.e., for using VSL figures in different contexts. We

therefore wish to see if the coefficients on the causes of death remain significant even after we include all of these alternative- and beneficiary-specific variables.

6. The Data.

We interviewed a total of 983 respondents in Milan, Italy.¹³ Descriptive statistics of this sample are reported in table 2. As per our sampling plan, we have a roughly even number of men and women, and the respondents are uniformly distributed in the 20-34, 35-44 and 45-60 age groups. The average age is 39. The sample is in line with the population of Milan for education and income. Mean (after tax) household income is about \in 30,000 a year.

The respondents are generally in good health. About 53% of them indicated that they were in "excellent" or "very good" health compared to other people their age. About 18% of the respondents reported having a chronic respiratory illness (e.g., asthma or chronic bronchitis), and only 1.47% had or had previously had cancer.

Our questionnaire included several questions designed to test whether respondents understood the probability material that was presented to them. Only 7% failed question D1, which asked them to compute the probability of winning a lottery where 10,000 tickets were sold (and there is only one winning ticket). Fifteen percent failed question D2, which asked respondents to read a bar chart with the chance of dying over the next 5 years for children aged 0-4, young adults aged 25-29, and adults aged 40-44, and indicate which of these age groups had the highest chance of dying. Finally, question D3 checks basic numeracy with probabilities. If the chance of dying over the next 5 years for 20-24-year-olds is 30 in 10,000, how many deaths

¹³ An additional 923 respondents completed a variant of the questionnaire that was virtually identical, but where the beneficiary of the risk-reducing measures in the conjoint choice experiments was one of the respondent's children. We do not examine these latter 923 interviews in this paper.

do we expect to see in a population of 100,000 people in this age group? About 10.5% of the sample failed this quiz.

Since this questionnaire focuses on the cause of death, we asked people to rate the dread they associate with various causes of death on a scale from 1 to 5. The results of this exercise are reported in table 3. They show that the cancer scores very highly on the dread scale, with 56% of the respondents regarding a cancer death as "very high dread." Surprisingly, however, people did not have the same reaction to leukemia, although the latter is a form of cancer of the blood. Regarding the other causes of death examined in the conjoint choice experiments, 28% of the respondents considered road traffic accidents to be "high dread" and 12% assigned a similar rating to respiratory illnesses.

We also asked respondents to rate the effectiveness of private behaviors and actions, and public programs, respectively, in reducing the risk of dying for each of the three causes studied in this questionnaire. The results of this exercise are displayed in Figure 2. The figure shows that people are willing and capable of assessing the effectiveness of different ways of reducing mortality risks. It appears that people regard individual actions as somewhat more effective than public programs for road traffic risks and cancer, but this judgment is partially reversed for respiratory risks.¹⁴

Descriptive statistics about various measures of exposure, sensitivity and experience with the various risks are displayed in table 4. Table 4 confirms that people dread cancer more highly than the other causes of death. It also shows that, based on individual behaviors, "everyday" exposure is highest for road traffic risks, followed by respiratory illnesses and cancer. Most of

¹⁴ Almost 24% of the respondents consider public programs highly effective in reducing respiratory risks. Only 13.8% of the sample considers private behaviors very effective at reducing the risks of dying from respiratory illnesses.

the respondents consider road traffic risks larger than cancer risks. Based on health status, about 20% of the respondents are classified as "sensitive" to respiratory illnesses, and only 1.5% are "sensitive" to cancer risks. Experience with the cause of death—directly or via another person— is highest for cancer.

7. Estimation Results.

A. Basic Models

We fit a conditional logit corresponding to indirect utility (2), and estimate the VSL to be \in 4 million (standard error \in 0.257 million). Importantly, the estimation results (displayed in table 5, panel (A)) indicate clearly that the marginal utility of a risk reduction is positive and statistically significant, which confirm that the survey responses are well-behaved and pass a "scope" test.

When we account for the cause of death (i.e., we estimate the model of equation (7) with $a_4 = 0$; see table 5, panel (B)), we get VSL figures of €3.360 million (respiratory illness), €5.280 million (cancer) and €2.874 million (road traffic accidents). The discount rate is not statistically different from zero. Clearly, cancer mortality risk reductions are valued more highly than the other causes of death, an effect that may be due to high dread, high baseline risk, and/or low perceived controllability.¹⁵

B. The Effect of Risk Characteristics

¹⁵ For the sake of simplicity, we only display results for the runs where we assigned survival probabilities equal to those implicit in the mortality risks shown to the respondents in the survey. The results are robust to replacing them with the respondent-assessed probabilities, and to setting the probability of survival to 1. The results displayed in table 6 and 7 are robust to excluding from the sample those respondents who failed one or more of the probability quizzes.

Is accounting for these factors sufficient to explain away the differences in VSL? In table 6 we report the results of models where we add the variables described in section 5.B. The simplest of these models is specification (A) of table 6, where we simply enter an interaction between DR and a public program dummy (see equation (7)). The coefficient on this variable is positive and significant, and implies that the VSL for any given cause of death is $\notin 0.918$ million greater when the risk reduction is delivered by a public program.

We check for possible heterogeneity in this effect by entering two interactions in run (B), namely whether the risk reduction comes via a public program interacted with respondent rating of the effectiveness of public programs at reducing that type of risk, and private behavior times the effectiveness rating of private behaviors.¹⁶ The result of adding these variables is striking. First, the coefficient on public program is dramatically larger than in specification (A). Second, both effectiveness measures are positively and significantly associated with the VSL. Third, respondents seem to treat the effectiveness of public and private programs asymmetrically and the impact of the former is almost twice as large as that of the latter.

In (C) and (D), we check whether the value of a unit of risk reduction depends on the baseline risk of dying (for all causes), assuming that respondents accept the average mortality risks for people their age shown to them in the questionnaire. Whether DR is interacted with baseline risks (model (C)), log baseline risks (run (D)), or dummies for each baseline risk level (not reported), the results are the same: The VSL does not depend systematically on total baseline risks.

The model in (E) enters interactions between DR and a researcher-constructed measure of exposure (EXPOSURE), the respondent's assessment about the magnitude of the risks of dying

¹⁶ Both of these interactions are further multiplied by the size of the risk reductions.

for specific causes among people his or her age (MORECOMMON), whether the respondent has chronic respiratory illnesses or cancer, which would raise their risks of dying for the indicated causes (SENSITIVITY), and EXPERIENCE with each of the three causes of death. Baseline risk remains insignificant, but at least two of these newly added variables (EXPOSURE and EXPERIENCE) do have significant additional explanatory power. With the exception of SENSITIVITY, all of them enter in the model with a positive coefficient.

Specifically, for people who are highly exposed to the risk being valued the VSL is $\notin 0.422$ million higher, and those with experience with the risk being valued have VSL figures that are $\notin 0.516$ million larger than those with no experience. When the risk being valued is thought to be more common for people one's age, the corresponding VSL is about $\notin 0.394$ million larger, but this effect is significant only at the 10% level.

We add cause-specific dread ratings in (F), along with a dummy denoting whether the respondent reported the highest level of dread for this particular risk. Doing so reduces slightly the coefficients on exposure, common cause of death, and experience. Even more importantly, dread enters positively and significantly in the regression. Each additional dread point increases the VSL by $\notin 0.322$ million. If a cause of death received the highest dread score, its VSL is an additional $\notin 0.402$ million greater. The effect of the highest dread dummy is large, but not statistically significant at the conventional levels.

Comparison across models (A)-(F) shows that the effects of the main attributes of risk are robust to the changes in specification. Moreover, there is evidence of considerable heterogeneity in valuation, since as we add additional controls the effect of a risk reduction per se (i.e., α_1) first vanishes and then is replaced by the effects of the risk interactions. Importantly, models (A)-(F) show that while exposure, sensitivity, experience with the risk in question and one's dread of it

affect the VSL and indeed account for a sizeable portion of it, reductions in cancer mortality risks are valued *more* than a respiratory risk reduction of comparable size and similar characteristics, including dread rating.

To get a sense of the contribution of each regressor to the VSL, we begin with using specification (F) to compute the respiratory-illness VSL for a 35-39 person who is in good health, has no exposure or experience with this risk, examines a private risk reduction, rates the latter's effectiveness a 3, and has the average dread (3.758) for respiratory illnesses. For this respondent, the VSL is $\in 0.382$ million. Increasing the effectiveness to 4 raises the VSL by $\notin 0.248$ million, bringing it to $\notin 0.630$ million (a 65% increase).

Changing the provision to a public program brings the VSL to $\notin 2.070$ million for the lowest level of effectiveness (1), and each additional point of public-program effectiveness raises the VSL by $\notin 0.494$ million. Suppose now that the dread score was 5, holding the effectiveness at 2. This would further raise the VSL to $\notin 3.366$ million. Changing the cause of death to cancer (which typically elicits high dread ratings, so we leave the dread score at 5) while holding all else the same would raise the VSL to $\notin 5.866$ million. In other words, the cancer VSL is worth $\notin 0.952$ million *more* than a respiratory risk reduction with otherwise similar characteristics. By contrast, the VSL for road traffic risks is about $\notin 1.214$ million *less* than that for respiratory risk, all else the same.

The calculations show that the public program attribute has a larger effect on the VSL ($\notin 2$ million) than cancer (a little less than one million), and that changing the belief in the effectiveness of a public from 1 to 5 increase the VSL by almost as much ($\notin 1.976$ million). Changing the level of dread from 1 to 5 increases the VSL by roughly the same amount ($\notin 2.012$ million).

C. The Effect of Individual Characteristics

Do individual characteristics of the respondent further account for heterogeneity in mortality risk valuation? We investigate this question in table 7. We restrict attention to gender, education and income.¹⁷ The results of panel (A) of table 7 suggest that the VSL increases with the respondent education, although not in a monotonic fashion (it is higher for those with a few years of college than for those with a college degree), and that is lower among women. Since our data indicate that dread of any given cause of death is actually higher among women, we suspect it might be due to lower incomes or women's reluctance to commit to spending money for themselves without checking with other family members first.¹⁸

In panel (B) of table 7 we allow the marginal utility of income to be different across people whose household income is above or below the citywide average (\notin 30,000 a year). Those with household income below the average indeed have a marginal utility of income that is about 20% higher than the others, an effect that is consistent with the predictions of economic theory.¹⁹ This decreases the marginal values of risk reductions by 20% among lower-income respondents, but does not explain away why women should be willing to pay less than men, all else the same.²⁰

In sum, there is evidence that certain individual characteristics such as gender, education and income do affect the VSL. Including these variables, however, does not change the marginal

¹⁷ Much attention has been devoted as of late to the relationship between the VSL and age to empirically check whether the elderly (who are often the primary beneficiaries of environmental policies) have a lower VSL. Earlier work (summarized in Krupnick, 2007) finds that the decline in the VSL, if any, is observed primarily among individuals aged 70 and older. We do not interview individuals over age 60, and for this reason we found little evidence of an association between age and the VSL. In runs not reported, we also entered interactions between the risk reduction and age dummies, but the results were ambiguous, and the relationship with the VSL non-monotonic and non-quadratic.

¹⁸ Alberini et al. (2007) report a similar effect in an earlier survey on mortality risks in Italy.

¹⁹ This is calculated by summing coefficients β and β_2 .

²⁰ We also estimated a model where the marginal utility of income is a linear function of income, but did not find the coefficient on income to be significant.

effect of other attributes or risk perceptions on the VSL, and does not change the result that, all else the same, the cancer VSL is higher than the respiratory-illness VSL, which, in turn, is greater than the VSL in the road-traffic risks.

D. Robustness Checks

In table 8, run (A), we enter interactions of DR with variables that further capture the controllability of risks and the ability of the respondent to identify triggers of these risks. These measures of controllability are AUTOAVD×DR×whether the risk is for road traffic accidents and CANCCOMM×DR×whether the risk is cancer, where AUTOAVD is a dummy equal to one if the respondents agrees very strongly that it is possible to avoid most road traffic accidents, and CANCCOMM represents agreement (with strength 4 or 5) with the statement that "cancer is so common that in most family someone will get cancer."

The respondent's ability to identify causes of disease and death are captured by MANYYRS, a dummy equal to one if the respondent agrees with strength 4 or 5 that it takes exposure to carcinogens for many years before one develops cancer, and POLLRESP, which is a equal to one if the respondent agrees strongly with the statement that the only consequence of exposure to air pollution is that someone will develop respiratory problems. These variables are positively associated with the likelihood of choosing an alternative, but the only one that is statistically significant is whether one thinks that cancer will become more and more common.

In (B), we include in the model an alternative-specific intercept for the status quo, which has is negative and significant, implying that people choose the status quo less frequently than is predicted by a model where the status quo is simply obtained by coding all attributes to zero. Including this variable changes the estimates of a_1 but leaves most other coefficients virtually the same. The VSL for a specified risk reduction profile changes, but the qualitative results about the influence of various factors on the VSL hold. The additional effect of cancer on the VSL is $\in 0.850$ million more than that predicted by the other attributes of risk, and that of road accidents $\in 0.780$ million less.

One concern with our regressions is that there may be an element of arbitrariness in the construction of the regressors. For this reason, in (C) we experiment with alternative definitions of the exposure, sensitivity, and experience variables, as described in the Appendix. We find that the results are virtually unchanged when these new definitions are used: The coefficient estimates in panel (C) are very close to their counterparts in panel (F) of table 6.

8. Conclusions.

We have used conjoint choice experiments, which we administered to a sample of persons aged 20-60 in Milan, Italy, to study whether the VSL—the WTP for a unit reduction in the risk of dying—varies with the cause of death, the mode of provision of the risk reduction, the futurity of the risk. We focused on three types of risk of dying—for respiratory illnesses, cancer, and in road-traffic accidents. We have attempted to see if the "cause of death" per se retains explanatory power for the responses to the conjoint choice questions after we account for subjective risk perceptions, including dread, exposure, controllability (measured by the perceived effectiveness of public and private measures), perceived baseline risk, and sensitivity to risk, which we measure with the presence or absence of the relevant chronic health conditions. We also control for experience with the risk in question.

The results of this investigation are striking. We find that the responses to the conjoint choice questions exhibit a remarkable degree of internal validity. The VSL increases with dread,

our construct of the respondent's exposure to the three types of risk, and perceived cause-specific baseline risks, and experience with the risk.

Perhaps the most striking result is that the VSL is much higher when the risk reduction is delivered by a public program, and even higher when the respondent believes that a government program is likely to be effective in reducing the indicated risk. Importantly, the respondent's assessment of the effectiveness of a private behavior in reducing risk is positively associated with the VSL, but the marginal effect of a one-point change in the effectiveness score is only half as much as public program effectiveness.

Even more important, despite carefully characterizing the mortality risk reduction using the above mentioned variables, the cause of death per se—namely, whether it's cancer, a roadtraffic accident or a respiratory illness—has an additional effect on the VSL. All else the same (including the degree of dread and controllability that the respondent ascribes to this risk), the fact that the cause of the death is cancer results in a VSL that is just under one million euro *above* what is predicted by dread, exposure, beliefs, etc. The VSL in the road traffic accident context is a just over one million euro *less* than what is predicted by dread, exposure, beliefs, etc.

Clearly, this has potentially important policy implications, and raises the question whether the lives saved by environmental regulations should be valued differently, depending on what the cause of death is. In the case of air pollution, which has been linked with increased risk of dying for cancer, cardiovascular disease, and respiratory disease, this issue appears to be particularly important. Currently, in its policy analyses, the European Commission applies a 50% cancer premium to the "general" VSL. This practice is consistent with our findings.

The magnitude of these effects, however, is smaller than that of the public program alone, that of public program effectiveness, and that of dread. This raises the question whether relying

on the VSL estimated from private risk reduction, as is often done in an effort to avoid double contain, may seriously understate society's willingness to pay for safety improvements brought by regulations and environmental programs.

References

- Alberini, Anna (2005), "What is a Life Worth? Robustness of VSL Values from Contingent Valuation Surveys," *Risk Analysis*, 25(4), 183-800.
- Alberini, Anna and Aline Chiabai (2007a), "Urban Environmental Health and Sensitive Populations: How Much are the Italians Willing to Pay to Reduce their Risks?" *Regional Science and Urban Economics*, 37(2), 239-258.
- Alberini, Anna and Aline Chiabai (2007b), "Discount Rates in Risk versus Money and Risk versus Risk Tradeoffs" *Risk Analysis*, 27(2), 483–498.
- Alberini, Anna, Stefania Tonin, Margherita Turvani, and Aline Chiabai (2007), "Paying for Permanence: Public Preferences for Contaminated Site Cleanup," *Journal of Risk and Uncertainty*, 34(2), 155-178.
- Aldy, Joseph E. and W. Kip Viscusi (2007), "Age Differences in the Value of Statistical Life: Revealed Preference Evidence," *Review of Environmental Economics and Policy*, 1, 241-260.
- Bateman, Ian et al. (2002), *Economic Valuation with Stated Preference Techniques: A Manual*, Cheltenham, UK: Edward Elgar.
- Bateman, Ian J., Diane Burgess, W. George Hutchinson, and David I. Matthews (2008), "Learning Design Contingent Valuation (LDCV): NOAA Guidelines, Preference Learning and Coherent Arbitrariness," *Journal of Environmental Economics and Management*, 55(2), 127-141.
- Bhattacharya, Soma, Anna Alberini and Maureen L. Cropper (2007), "The Value of Mortality Risk Reductions in Delhi, India," *Journal of Risk and Uncertainty*, 34(1), 21-47.
- Chauvin, Bruno, Daniele Hermand, and Etienne Mullet (2007), "Risk Perception and Personality Facets," *Risk Analysis* **27**(1), 171-185.
- Chilton, Susan, Judith Covey, Lorraine Hopkins, Michael Jones-Lee and Graham Loomes, et al. (2002), "Public Perceptions of Risk and Preference-Based Values of Safety," *Journal of Risk and Uncertainty*, 25(3), 211-232.
- Chilton, Susan, Michael Jones-Lee, Francis Kiraly, Hugh Metcalf, and Wei Pang (2006), "Dread Risks," *Journal of Risk and Uncertainty*, 33(3),165-182.
- Corso, Phaedra S., James K. Hammitt, John D. Graham (2001), "Valuing Mortality-Risk Reductions: Using Visual Aids to Improve the Validity of Contingent Valuation," *Journal of Risk and Uncertainty*, 23(2), 165-184.
- Cropper, Maureen, and Frances Sussman (1990), "Valuing Future Risks to Life," Journal of Environmental Economics and Management, 19, 160-174.

- Dickie, Mark and Shelby Gerking (1996), "Formation of Risk Beliefs, Joint Production and Willingness to Pay to Avoid Skin Cancer," *Review of Economics and Statistics*, 78(3), 451-463.
- Dickie, Mark and Shelby Gerking (2006), "Valuing Children's Health: Parental Perspectives," Chapter 3 in OECD, *Economic Valuation of Environmental Risks to Children*, Paris: OECD Publishing.
- Dockins, C., Kelly, Maguire, Nathalie Simon, and Melonie Sullivan (2004), "Value of Statistical Life Analysis and Environmental Policy: A White Paper," US Environmental Protection Agency EE-0483-01, Washington, DC, available at <u>http://yosemite.epa.gov/ee/epa/eermfile.nsf/vwAN/EE-0483-01.pdf/\$File/EE-0483-01.pdf</u>.
- Eeckhoudt, L. R., and J. K. Hammitt (2001), "Background Risks and the Value of a Statistical Life," *Journal of Risk and Uncertainty*, Vol. 23(3), 261-79.
- Evans, Mary F. and V. Kerry Smith (2006), "Do We Really Understand the Age-VSL Relationship?" *Resource and Energy Economics*, 28, 242-261.
- Fischhoff, Baruch, Paul Slovic, Sarah Lichtenstein, Stephen Read and Barbara Combs (1978), "How Safe is Safe Enough? A Psychometric Study of Attitudes towards Technological Risks and Benefits," *Policy Sciences* 9, 127–152.
- Hammitt, James K. and Jin-Tan Liu (2004) "Effects of Disease Type and Latency on the Value of Mortality Risk," *Journal of Risk and Uncertainty*, 28(1), 73-95.
- Hammitt, James K. and Kevin Haninger (2010), "Valuing fatal risks to children and adults: Effects of Disease, Latency, and Risk Aversion," *Journal of Risk and Uncertainty*, 40, 57– 83.
- Hurley, Fintan, Alistair Hunt, Hilary Cowie, Mike Holland, Brian Miller, Stephen Pye, and Paul Watkiss (2005), "Methodology for the Cost-Benefit Analysis for CAFE: Volume 2: Health Impact Assessment," AEAT/ED51014/Methodology Volume 2 Issue 1, February, available <u>http://ec.europa.eu/environment/archives/air/cafe/pdf/cba_methodology_vol2.pdf</u> (accessed 18 Jan 2010)
- Itaoka, Kenshi, Aya Saito, Alan Krupnick, Wiktor Adamowicz and Taketoshi Taniguchi (2006), "The Effect of Risk Characteristics on the Willingness to Pay for Mortality Risk Reductions from Electric Power Generation," *Environmental and Resource Economics*, 33, 371-398.
- Johannesson, Magnus, and Per-Olov Johansson (1996a), "To Be, or Not to Be, That Is the Question: An Empirical Study of the WTP for an Increased Life Expectancy at an Advanced Age," *Journal of Risk and Uncertainty*, 13, 163-174.

- Johannesson, Magnus, Per-Olov Johansson and Karl-Gustav Lofgren (1997), "On the Value of Changes in Life Expectancy: Blips Versus Parametric Changes," *Journal of Risk and Uncertainty*, 15(3), 221-239.
- Johannesson, Magnus, Per-Olov Johansson and Richard M. O'Conor (1996b), "The Value of Private Safety Versus the Value of Public Safety," *Journal of Risk and Uncertainty*, 13, 263-275.
- Johannesson, M. and D. Meltzer (1998) 'Some Reflections on Cost-Effectiveness Analysis' *Health Economics* 7: 1-7.
- Jones-Lee, M. W., M. Hammerton, and P. R. Phillips (1985), "The Value of Safety: Results of a National Sample Survey," *The Economic Journal*, 95, 49-72.
- Jones-Lee, M. W. (1989), The Value of Safety and Physical Risk, Blackwell, Oxford.
- Jones-Lee, M.W. (1991), "Altruism and the Value of Other People's Safety", *Journal of Risk* and Uncertainty, 4, 213-19.
- Jones-Lee, M.W. (1992), "Paternalistic Altruism and the Value of Statistical Life", *Economic Journal*, 102, 80-90.
- Krupnick, Alan (2007), "Mortality-risk Valuation and Age: Stated Preference Evidence," *Review* of Environmental Economics and Policy, 1, 261-282
- Krupnick, Alan, Anna Alberini, Maureen Cropper, Nathalie Simon, Bernie O'Brien, Ron Goeree, and Martin Heintzelman (2002), "Age, Health and the Willingness to Pay for Mortality Risk Reductions: A Contingent Valuation Study of Ontario Residents," *Journal* of Risk Uncertainty, 24, 161-186.
- Magat, Wesley A., W. Kip Viscusi, Joel Huber, (1996), "A Reference Lottery Metric for Valuing Health," *Management Science*, 42(8), 1118-1130.
- McDaniels, Timothy L., Mark S. Kamlet, and Gregory W. Fischer (1992), "Risk Perception and the Value of Safety," *Risk Analysis*, 12, 495-503.
- Mendeloff, John M. and Robert M. Kaplan (1989), "Are Large Differences in "Lifesaving" Costs Justified? A Psychometric Study of the Relative Value Placed on Preventing Deaths," *Risk Analysis* 9(3), 349–363.
- National Academy of Science (2008), "Estimating Mortality Risk Reduction and Economic Benefits from Controlling Ozone Air Pollution," Washington, DC, available at <u>http://www.nap.edu/catalog.12198.html</u>.
- Nielsen, J. Seested, Susan Chilton, Michael Jones-Lee, and Hugh Metcalf (2009), "How Would You Like Your Gain in Life Expectancy to be Provided? An Experimental Approach," Newcastle Discussion Paper in Economics 2009/09, University of Newcastle.

- Persson, Ulf, Anna Norinder, Krister Hjalte, Katarina Gralen (2001), "The Value of a Statistical Life in Transport: Findings from a New Contingent Valuation Study in Sweden", *The Journal of Risk and Uncertainty*, 23(2), 121–134.
- Pratt, John W. and Richard J. Zeckhauser (1996), "Willingness to Pay and the Distribution of Risk and Wealth," *Journal of Political Economy*, 104(4), 747-763.
- Revesz, Richard L. (1999), "Environmental Regulation, Cost Benefit Analysis, and the Discounting of Human Lives," *Columbia Law Review*, 194(4), 941-1017.
- Rowlatt, Penelope, Michael Spackman, Sion Jones, Michael Jones-Lee, and Graham Loomes (1998), "Valuation of Deaths from Air Pollution," National Economic Research Associates, London.
- Savage, Ian (1993), "An Empirical Investigation into the Effect of Psychological Perception on the Willingness-to-Pay to Reduce Risk," *Journal of Risk and Uncertainty*, 6, 75–90.
- Slovic, Paul (1987), "Perception of Risk," Science, 236(4799), 280-285.
- Starr, Chauncey (1969), "Social Benefit versus Technological Risk. What is Our Society Willing to Pay for Safety?" *Science*, 165(3899), 1232-1238.
- Sunstein, Cass R. (2004), "Value of Life: A Plea for Disaggregation," *Duke Law Journal*, 54, 385-444.
- Tolley, George S., Donald Kenkel, Robert Fabian (Eds.) (1994), Valuing Health for Policy: An Economic Approach, Chicago: University of Chicago Press.
- Tonin, Stefania, Anna Alberini, and Margherita Turvani (2009), "The Value of Reducing cancer Risks at Contaminated Sites: Are More Heavily Exposed People Willing to Pay More?" FEEM working paper 09.60, Milan, Italy, July.
- Tsuge T., A. Kishimoto, K. Takeuchi (2005), "A Choice Experiment Approach to the Valuation of Mortality", *The Journal of Risk and Uncertainty*, 31(1), 73–95.
- Urban, Jan, and Milan Ščasný (2007), "Determinants of Risk Perception Bias: An Empirical Study of Economically Active Population of the Czech Republic," Paper presented at the Conference "World of Labour and Quality of Life in Globalized Economy," University of Economics at Prague, September 2007.
- US Environmental Protection Agency (1999a), The Benefits and Costs of the Clean Air Act, 1990-2010, EPA Report to Congress, EPA-410-R-99-001. Office of Air and Radiation. Office of Policy, US EPA, Washington, DC, November. Available at <u>http://www.epa.gov/air/sect812/1990-2010/fullrept.pdf</u> (accessed 3 January 2010)
- US Environmental Protection Agency (1999b), Final Tier 2 Rule: Air Quality Estimation, Selected Health and Welfare Benefits Methods, and Benefits Analysis Results, EPA-420-R-99-032. Office of Air Quality Planning and Standards, Research Triangle Park, NC,

December. Available at <u>http://www.epa.gov/tier2/frm/tsd/r99032.pdf</u> (accessed 3 January 2010).

- US Environmental Protection Agency (2000), Guidelines for Preparing Economic Analyses, EPA-R-00-003, Washington, DC, September.
- Van Houtven, George, Melonie B. Sullivan, Chris Dockins (2008), "Cancer Premiums and Latency Effects: A risk Tradeoff Approach for Valuing Reductions in Fatal Cancer Risks," *Journal of Risk and Uncertainty*, 36, 179-199.
- Viscusi, W. Kip (1993), "The Value of Risks to Life and Health," *Journal of Economic Literature*, 31(4), 1912-1946.
- Viscusi W. Kip and Joseph E. Aldy (2003), "The Value of a Statistical Life: A Critical Review of Market Estimates Throughout the World," *Journal of Risk and Uncertainty*, 27(1), p. 5-76.

C La salute neg	gli adulti e bambini: e	quali sono le tue opinioni? - Windows Internet Explo	rer						
C 🖸 🗸 🖻	http://localhost/verhi/sur	vey.php		Google	P - @				
File Edit Vie	ew Favorites Tools H	telp							
Google G-		🖌 Go 🗄 🍏 🧭 🚰 👻 🗢 🤹 🖌 🏠 Bookmarks 🕶 🔯 637 b	locked 🛛 🍄 Check 👻 🔦 AutoLink 👻 🔚	AutoFill 🍙 Send to 🗸 🖉	Settings 7				
🚖 🏟 🍘 La	salute negli adulti e bambir	ii: quali sono le tue opini		🟠 🔹 🔝 👘 👻 Pag	je 🕶 🎯 Tools 🔹 🎽				
	La salute negli a	adulti e bambini: quali sono le tue opinioni?	5	Sezione I - Esercizi di scelta	<u>^</u>				
	Confronto sesso, stat sotto.	2 - Consideriamo altri due interventi che agis- to di salute, stile di vita e azioni preventive, attu Commerciatione	cono sulla tua probabilità di mori almente pari a 57 su 10.000 nel	re per una persona della tua età, corso di 5 anni, come descritti qui					
		Caratteristiche							
		Tipo di iniziativa	Azione preventiva individuale	Programma governativo nazionale					
		Altri beneficiari della riduzione della probabilità di morire	No	Altri adulti					
		Riduzione della probabilità di morire	2 su 10.000 in 5 anni	3 su 10.000 in 5 anni					
		Quando inizia la riduzione della probabilità di morire?	Tra 2 anni	Tra 2 anni					
		Costo per il tuo nucleo familiare, da pagarsi immediatamente e in un'unica soluzione	200 euro	2.000 euro					
	2. Quale sceglieresti fra le seguenti alternative, l'intervento C, l'intervento D, o nessuno dei due? O Intervento C Intervento D Nessuno dei due (preferisco non ottenere nessuna delle due riduzioni della probabilità di morire e non pagare nessuna delle due cifre)								
🔔 Done	11-11-11-11-11-11-11-11-11-11-11-11-11-			S Local intranet	€ 100% ·				
🎒 start	66 🖸 🗎	🖉 L. 😌 Search Desktop		<u></u>	🕑 🗞 🚺 🔝 10:37 PM				

Figure 1. Example of a Conjoint Choice Experiment question (for TFORMAT=2).

Figure 2. Respondent rating of the effectiveness of behaviors and public programs in reducing mortality risks.



Attribute	No. levels	Levels
Context (cause of death)	3	Cancer road traffic accidents respiratory illnesses
Private good or public program	2	private good (no other beneficiaries); nationwide public program (other beneficiaries)
Latency	4	0, 2, 5, 10 years
Size of the risk reduction	4	2, 3, 5, 7 in 10,000 over 5 years
(one-time) Cost to the respondent	4	200, 500, 1000, 2000 euro

Table 1. Summary of attributes and attribute levels in the conjoint choice experiments.

Variable	Description	Ν	Mean	Std.	Min.	Max.
				dev.		
age	age of the respondent	983	39.211	9.965	20	60
female	dummy=1 if female	983	0.505	0.5	0	1
single	dummy=1 if single	983	0.077	0.267	0	1
married	dummy=1 if married	983	0.879	0.326	0	1
income	annual net household income in euro	974	30,008	11,371	5,000	80,000
homemaker	dummy=1 if homemaker	983	0.07	0.256	0	1
Matura	dummy=1 if high school diploma	983	0.4	0.49	0	1
somecoll	dummy=1 if some years at university without degree	983	0.052	0.222	0	1
College	dummy=1 if college (university) degree	983	0.25	0.433	0	1

Table 2.	Descriptive	statistics	of the	sample.

	Percent of the respondents						
	no dread	m	edium dread		high dread		
	1	2	3	4	5		
road traffic accident	2.86	7.96	27.55	33.78	27.86		
leukemia	7.16	18.94	21.19	25.28	27.43		
cardiovascular disease	3.58	13.82	31.93	35.62	15.05		
chronic respiratory illnesses	7.26	18.92	33.95	27.61	12.27		
cancer	2.35	3.99	13.61	24.46	55.58		
fire	14.31	25.97	23.21	18.51	18.00		

Table 3. Dread ratings assigned by the respondents to various causes of death.

Table 4. Exposure and sensitivity variables.

Variable name	Variable type	ROAD TRAFFIC ACCIDENT	RESPIRATORY PROBLEMS	CANCER
DREAD	scale (1 to 5)	3.758	3.187	4.269
HIDREAD	dummy	0.279	0.123	0.556
EXPOSURE	dummy	0.852	0.795	0.494
MORECOMMON	dummy	0.710	0	0.511
SENSITIVITY	dummy	0	0.204	0.016
EXPERIENCE	dummy	0.119	0.044	0.518
CONTROLLABILITY*	dummy	0.280	0	0.427 (CANCCOMM) & 0.306 (MANYYRS)

* For road traffic accidents, dummy AUTOAVD=1 if the respondent agrees strongly that "most road traffic accidents could be avoided." For cancer, MANYYRS=1 if the respondent agrees strongly or very strongly that "it takes exposure to carcinogens for many years before developing cancer," CANCCOMM=1 if respondent agrees strongly or very strongly that "cancer is so common that in virtually all families someone will get cancer."

	(/ Basic I	A) Model	(B) Controlling for cause of death		
	Coefficient	t stat	Coefficient	t stat	
α1	0.1687	16.899	0.1539	14.881	
lpha2 (cancer)			0.0880	8.152	
lpha3 (road traffic)			-0.0223	-2.483	
β	-0.0004	-16.112	-0.0005	-17.042	
δ (discount rate)	-0.0052	-0.614	-0.0062	-0.766	
	VSL (mill. euro)	Std. err. around VSL	VSL (mill. euro)	Std. err. around VSL	
All causes	4.000	0.257			
Respiratory			3.360	0.222	
Cancer			5.281	0.346	
Road traffic			2.874	0.231	
Ν	7261		7261		
mean log L	-6694.11		-6628.42		

Table 5. Estimation results: Simplest Models. No dominated pair.

47	

	(4	4)	(1	3)	(C)	([)	(6	E)	(F)
	coeff.	t stat	coeff	t stat	coeff	t stat	coeff	t stat	coeff	t stat	coeff	t stat
α1	0.1223	11.535	-0.0285	-1.019	-0.0362	-1.262	-0.0305	-0.74	-0.0431	-1.023	-0.0774	-1.705
α 2 (cancer)	0.0853	8.393	0.0841	8.122	0.0842	8.125	0.0841	8.121	0.0693	5.176	0.0476	3.425
lpha3 (road traffic)	-0.0247	-2.878	-0.0342	-3.859	-0.0343	-3.867	-0.0342	-3.859	-0.0514	-4.14	-0.0607	-4.784
α 4 (PUBLIC)	0.0459	7.007	0.1046	4.129	0.1038	4.095	0.1045	4.127	0.1004	3.947	0.0969	3.787
α 5 PUBEFF			0.0273	5.622	0.0276	5.657	0.0273	5.622	0.0268	5.494	0.0247	5.025
α 6 PRIVEFF			0.0143	2.252	0.0141	2.213	0.0143	2.25	0.0134	2.095	0.0124	1.928
α 7 BRISK					0.0074	1.223			• • •		* * *	
α 8 ln(BRISK)					-		0.0005	0.068	0.0007	0.094	-0.0003	-0.039
α 9 EXPOSURE					-				0.0211	2.073	0.0177	1.726
α 10 MORECOMM					-				0.0197	1.816	0.0161	1.479
α 11 SENSITIVITY					-				-0.0009	-0.054	-0.0049	-0.288
α 12 EXPERIENCE					-				0.0258	2.144	0.0215	1.766
α 13 DREAD					-						0.0161	2.662
α 14 HIDREAD					-				• • •		0.0201	1.388
					- - - -				5 5 6 7		* * * *	
β	-0.0005	-16.819	-0.0005	-16.64	-0.0005	-16.656	-0.0005	-16.64	-0.0005	-16.675	-0.0005	-16.646
δ	-0.0148	-1.932	-0.0163	-2.173	-0.0161	-2.149	-0.0163	-2.171	-0.0156	-2.089	-0.015	-2.017
					-				- - - -		• • • •	
Ν	7261		6970		6970		6970		6970		6970	
log L	-6603.65		-6304		-6303.24		-6304		-6296.27		-6282.04	

Table 6. Estimation results: Effect of Exposure and Risk Perceptions. No dominated pair.

	(A)		(B)	
	coeff.	t stat	coeff	t stat
α1	-0.0655	-1.39	-0.0572	-1.21
lpha2 (cancer)	0.0435	3.149	0.0428	3.096
lpha3 (road traffic)	-0.0612	-4.845	-0.0616	-4.877
α4 (PUBLIC)	0.0987	3.883	0.0993	3.907
α 5 PUBEFF	0.0255	5.191	0.025	5.083
α 6 PRIVEFF	0.0128	2.008	0.0132	2.064
α 7 ln(BRISK)	0.0005	0.076	-0.0001	-0.016
$\alpha 8 \text{ EXPOSURE}$	0.0175	1.716	0.0186	1.827
α 9 MORECOMM	0.0173	1.597	0.0165	1.522
α 10 SENSITIVITY	-0.0064	-0.378	-0.0103	-0.603
α 11 EXPERIENCE	0.0223	1.849	0.0208	1.728
α 12 DREAD	0.0158	2.638	0.0153	2.556
α 13 HIDREAD	0.0229	1.595	0.0236	1.65
α 14 MATURA	-0.0076	-0.558	-0.0093	-0.682
α 15 SOMECOLL	0.0528	1.878	0.0506	1.803
α 16 COLLEGE	0.0249	1.577	0.0203	1.283
α 17 FEMALE	-0.0505	-4.389	-0.0513	-4.455
β	-0.0005	-16.617	-0.0005	-14.745
δ	-0.0167	-2.263	-0.0174	-2.349
β_2 (dummy if income			0.0001	2 224
below life average)			0.0001	2.004
Ν	6965		6910	
loa L	-6261.35		-6204.88	

 Table 7. Estimation results: Effect of the Individual Characteristics of the Respondents. No dominated pair.

		4)	(E	3)	(C)	
	coeff	t stat	coeff	t stat	coeff.	t stat.
α1	-0.0582	-1.222	-0.1623	-3.201	-0.084	-1.86
lpha2 (cancer)	0.0405	2.414	0.051	3.054	0.0641	4.444
lpha3 (road traffic)	-0.0554	-3.646	-0.0471	-3.036	-0.0608	-4.926
α 4 (PUBLIC)	0.0988	3.875	0.1053	4.101	0.0972	3.806
α 5 PUBEFF	0.0244	4.932	0.0256	5.005	0.0246	5.016
α 6 PRIVEFF	0.0131	2.035	0.014	2.196	0.0124	1.934
lpha7 ln(BRISK)	-0.0007	-0.097	-0.0024	-0.314	-0.0005	-0.068
α 8 EXPOSURE	0.016	1.561	0.0161	1.584	0.0255	2.382
α 9 MORECOMM	0.0152	1.4	0.0161	1.489	0.0173	1.597
α 10 SENSITIVITY	-0.0123	-0.718	-0.0122	-0.718	-0.003	-0.067
α 11 EXPERIENCE	0.0169	1.385	0.0188	1.559	0.0036	0.336
α 12 DREAD	0.0152	2.524	0.0151	2.484	0.0166	2.747
α 13 HIDREAD	0.0215	1.492	0.0198	1.39	0.0198	1.375
α 14 MATURA	-0.0091	-0.666	-0.0125	-0.88		
α 15 SOMECOLL	0.05	1.776	0.0539	1.825		
α 16 COLLEGE	0.0209	1.317	0.0143	0.869		
α 17 FEMALE	-0.0514	-4.433	-0.059	-4.81		
α 18 CANCCOMM	0.0375	2.249	0.0374	2.254		
α 19 AUTOAVD	0.02	1.249	0.0197	1.247		
α 20 MANYYRS	0.002	0.115	0.0002	0.01		
α 21 POLLRESP	0.0196	1.369	0.02	1.405		
β	-0.0005	-14.73	-0.0006	-17.428	-0.0005	-16.63
δ	-0.0168	-2.274	-0.0221	-1.886	-0.0155	-2.083
β_2 (dummy if income	0.0001	2 202	0.0001	2 501		
	0.0001	2.232	-0.9587	-16 827		
			-0.3007	-10.021		
N	6910		6910		6970	
log L	-6200.91		-6055.62		-6282.65	

Table 8. Estimation results: Additional checks. No dominated pair. Specification (C) uses alternative definitions for EXPOSURE and EXPERIENCE (see the Appendix).

Appendix

Table A.1: Alternative construction of variables to check for effects on the VSL

	Source	ROAD TRAFFIC ACCIDENT	RESPIRATORY PROBLEMS	CANCER
EXPOSURE(2)	constructed	driving at least once a week	Living in moderately or very polluted environment	cancer runs in respondent's family
SENSITIVITY(2)	respondent	n.a.	perception about bad actual or future health status	perception about bad actual or future health status
EXPERIENCE(2)	constructed	admitted to the hospital or taken to the emergency room in last 5 years for this reason	admitted to the hospital or taken to the emergency room in last 5 years for this reason, respondent has chronic respiratory illnesses.	close person or family member had cancer and respondent had cancer

Table A.2: Average values of factors used to explain VSL (see table A.1 for coding).

Variable name	Variable type	ROAD TRAFFIC ACCIDENT	RESPIRATORY PROBLEMS	CANCER
EXPOSURE(2)	dummy	0.852	0.748	0.204
SENSITIVITY(2)	dummy	0	0.020	0.020
EXPERIENCE(2)	dummy	0.119	0.224	0.685

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses:

http://www.feem.it/getpage.aspx?id=73&sez=Publications&padre=20&tab=1 http://papers.ssrn.com/sol3/JELJOUR_Results.cfm?form_name=journalbrowse&journal_id=266659

http://ideas.repec.org/s/fem/femwpa.html http://www.econis.eu/LNG=EN/FAM?PPN=505954494

http://ageconsearch.umn.edu/handle/35978

http://www.bepress.com/feem/

NOTE DI LAVORO PUBLISHED IN 2010

GC	1.2010	Cristina Cattaneo: <u>Migrants' International Transfers and Educational Expenditure: Empirical Evidence from</u>
		Albania
SD	2.2010	Fabio Antoniou, Panos Hatzipanayotou and Phoebe Koundouri: <u>Tradable Permits vs Ecological Dumping</u>
SD	3.2010	Fabio Antoniou, Panos Hatzipanayotou and Phoebe Koundouri: <u>Second Best Environmental Policies under</u>
SD	4 2010	Carlo Carraro, Enrica De Cian and Lea Nicita: Modeling Biased Technical Change Implications for Climate
50	4.2010	Policy
IM	5.2010	Luca Di Corato: Profit Sharing under the threat of Nationalization
SD	6.2010	Masako Ikefuji, Jun-ichi Itaya and Makoto Okamura: Optimal Emission Tax with Endogenous Location
		Choice of Duopolistic Firms
SD	7.2010	Michela Catenacci and Carlo Giupponi: Potentials and Limits of Bayesian Networks to Deal with
		Uncertainty in the Assessment of Climate Change Adaptation Policies
GC	8.2010	Paul Sarfo-Mensah and William Oduro: Changes in Beliefs and Perceptions about the Natural Environment
		<u>in the Forest-Savanna Transitional Zone of Ghana: The Influence of Religion</u>
IM	9.2010	Andrea Boitani, Marcella Nicolini and Carlo Scarpa: Do Competition and Ownership Matter? Evidence
		from Local Public Transport in Europe
SD	10.2010	Helen Ding and Paulo A.L.D. Nunes and Sonja Teelucksingh: <u>European Forests and Carbon Sequestration</u>
		Services : An Economic Assessment of Climate Change Impacts
GC	11.2010	Enrico Bertacchini, Walter Santagata and Giovanni Signorello: Loving Cultural Heritage Private Individual
	10.0010	Giving and Prosocial Behavior
SD	12.2010	Antoine Dechezleprêtre, Matthieu Glachant and Yann Ménière: <u>What Drives the International Transfer of</u>
C D	12 2010	Climate Change Mitigation Technologies? Empirical Evidence from Patent Data
SD	13.2010	Andrea Bastianin, Alice Favero and Emanuele Massetti: <u>Investments and Financial Flows Induced by</u>
CD	14 2010	Climate Mitigation Policies
SD	14.2010	Rever Genagn: <u>100 Much Oli</u>
	16.2010	Chiara Fumagaili and Massimo Motta: <u>A Simple Theory of Predation</u>
uc	10.2010	Rinaldo Drau, Adriana Di Liberto and Francesco Pigliaru: <u>Tourism and Development: A Recent</u>
SD	17 2010	<u>Prenomenon Built on Oid (Institutional) Roots:</u>
30	17.2010	MOSE Barriers on Venice Port Activities
SD	18 2010	ZhongXiang Zhang: Climate Change Meets Trade in Promoting Green Growth: Potential Conflicts and
50	10.2010	Svnergies
SD	19.2010	Elisa Lanzi and Ian Sue Wing: Capital Malleability and the Macroeconomic Costs of Climate Policy
IM	20.2010	Alberto Petrucci: Second-Best Optimal Taxation of Oil and Capital in a Small Open Economy
SD	21.2010	Enrica De Cian and Alice Favero: Fairness, Credibility and Effectiveness in the Copenhagen Accord: An
		Economic Assessment
SD	22.2010	Francesco Bosello: Adaptation, Mitigation and "Green" R&D to Combat Global Climate Change. Insights
		From an Empirical Integrated Assessment Exercise
IM	23.2010	Jean Tirole and Roland Bénabou: Individual and Corporate Social Responsibility
IM	24.2010	Cesare Dosi and Michele Moretto: Licences, "Use or Lose" Provisions and the Time of Investment
GC	25.2010	Andrés Rodríguez-Pose and Vassilis Tselios (lxxxvi): <u>Returns to Migration, Education, and Externalities in</u>
		the European Union
GC	26.2010	Klaus Desmet and Esteban Rossi-Hansberg (lxxxvi): <u>Spatial Development</u>
SD	27.2010	Massimiliano Mazzanti, Anna Montini and Francesco Nicolli: Waste Generation and Landfill Diversion
		Dynamics: Decentralised Management and Spatial Effects
SD	28.2010	Lucia Ceccato, Valentina Giannini and Carlo Gipponi: <u>A Participatory Approach to Assess the Effectiveness</u>
		of Responses to Cope with Flood Risk
SD	29.2010	Valentina Bosetti and David G. Victor: <u>Politics and Economics of Second-Best Regulation of Greenhouse</u>
		Gases: The Importance of Regulatory Credibility
IM	30.2010	Francesca Cornelli, Zbigniew Kominek and Alexander Ljungqvist: <u>Monitoring Managers: Does it Matter?</u>
GC	31.2010	Francesco D'Amuri and Juri Marcucci: "Google it!" Forecasting the US Unemployment Rate with a Google
CD.	22 2010	JOD Search Index
20	32.2010	Francesco Doseilo, Cario Carraro and Enrica De Cian: <u>Climate Policy and the Optimal Balance between</u>
		<u>willigation, Adaptation and Unavoided Damage</u>

SD	33.2010	Enrica De Cian and Massimo Tavoni: <u>The Role of International Carbon Offsets in a Second-best Climate</u>
SD	34.2010	ZhongXiang Zhang: The U.S. Proposed Carbon Tariffs, WTO Scrutiny and China's Responses
IM	35.2010	Vincenzo Denicolò and Piercarlo Zanchettin: Leadership Cycles
SD	36.2010	Stéphanie Monjon and Philippe Quirion: <u>How to Design a Border Adjustment for the European Union</u> Emissions Trading System?
SD	37.2010	Meriem Hamdi-Cherif, Céline Guivarch and Philippe Quirion: <u>Sectoral Targets for Developing Countries:</u> Combining "Common but Differentiated Responsibilities" with "Meaningful participation"
IM	38.2010	G. Andrew Karolyi and Rose C. Liao: <u>What is Different about Government-Controlled Acquirers in Cross</u> - Border Acquisitions?
GC	39,2010	Kietil Biorvatn and Alireza Naghavi: Rent Seekers in Rentier States: When Greed Brings Peace
GC	40.2010	Andrea Mantovani and Alireza Naghavi: Parallel Imports and Innovation in an Emerging Economy
SD	41.2010	Luke Brander, Andrea Ghermandi, Onno Kuik, Anil Markandya, Paulo A.L.D. Nunes, Marije Schaafsma and Alfred Wagtendonk: <u>Scaling up Ecosystem Services Values: Methodology, Applicability and a Case</u>
SD	42.2010	Valentina Bosetti, Carlo Carraro, Romain Duval and Massimo Tavoni: <u>What Should We Expect from</u> <u>Innovation? A Model-Based Assessment of the Environmental and Mitigation Cost Implications of Climate-</u>
		Related R&D
SD	43.2010	Frank Vöhringer, Alain Haurie, Dabo Guan, Maryse Labriet, Richard Loulou, Valentina Bosetti, Pryadarshi R. Shukla and Philippe Thalmann: <u>Reinforcing the EU Dialogue with Developing Countries on Climate</u>
GC	44.2010	Change Mitigation Angelo Antoci, Pier Luigi Sacco and Mauro Sodini: <u>Public Security vs. Private Self-Protection: Optimal</u>
IM	45 2010	<u>I axation and the Social Dynamics of Fear</u> Luca Enriques: European Takeover Law, The Case for a Neutral Approach
SD	46.2010	Maureen L. Cropper, Yi Jiang, Anna Alberini and Patrick Baur: Getting Cars Off the Road: The Cost-
02		Effectiveness of an Episodic Pollution Control Program
IM	47.2010	Thomas Hellman and Enrico Perotti: The Circulation of Ideas in Firms and Markets
IM	48.2010	James Dow and Enrico Perotti: <u>Resistance to Change</u>
SD	49.2010	Jaromir Kovarik, Friederike Mengel and José Gabriel Romero: (Anti-) Coordination in Networks
SD	50.2010	Helen Ding, Silvia Silvestri, Aline Chiabai and Paulo A.L.D. Nunes: <u>A Hybrid Approach to the Valuation of</u> <u>Climate Change Effects on Ecosystem Services: Evidence from the European Forests</u>
GC	51.2010	Pauline Grosjean (lxxxvii): <u>A History of Violence: Testing the 'Culture of Honor' in the US South</u>
GC	52.2010	Paolo Buonanno and Matteo M. Galizzi (Ixxxvii): <u>Advocatus, et non latro? Testing the Supplier-Induced-Demand Hypothesis for Italian Courts of Justice</u>
GC	53.2010	Gilat Levy and Ronny Razin (Ixxxvii): <u>Religious Organizations</u>
GC	54.2010	Matteo Cervellati and Paolo Vanin (lxxxvii): <u>"Thou shalt not covet": Prohibitions, Temptation and Moral Values</u>
GC	55.2010	Sebastian Galiani, Martín A. Rossi and Ernesto Schargrodsky (lxxxvii): <u>Conscription and Crime: Evidence</u> from the Argentine Draft Lottery
GC	56.2010	Alberto Alesina, Yann Algan, Pierre Cahuc and Paola Giuliano (Ixxxvii): <u>Family Values and the Regulation of</u> <u>Labor</u>
GC	57.2010	Raquel Fernández (lxxxvii): <u>Women's Rights and Development</u>
GC	58.2010	Tommaso Nannicini, Andrea Stella, Guido Tabellini, Ugo Troiano (Ixxxvii): <u>Social Capital and Political</u> <u>Accountability</u>
GC	59.2010	Eleonora Patacchini and Yves Zenou (Ixxxvii): Juvenile Delinquency and Conformism
GC	60.2010	Gani Aldashev, Imane Chaara, Jean-Philippe Platteau and Zaki Wahhaj (Ixxxvii): <u>Using the Law to Change the Custom</u>
GC	61.2010	Jeffrey Butler, Paola Giuliano and Luigi Guiso (Ixxxvii): <u>The Right Amount of Trust</u>
SD	62.2010	Valentina Bosetti, Carlo Carraio and Massimo Tavoni: <u>Alternative Paths toward a Low Carbon World</u>
SD	63.2010	Kelly C. de Bruin, Rob B. Dellink and Richard S.J. Tol: <u>International Cooperation on Climate Change</u> <u>Adaptation from an Economic Perspective</u>
IM	64.2010	Andrea Bigano, Ramon Arigoni Ortiz, Anil Markandya, Emanuela Menichetti and Roberta Pierfederici: <u>The</u> <u>Linkages between Energy Efficiency and Security of Energy Supply in Europe</u>
SD	65.2010	Anil Markandya and Wan-Jung Chou: <u>Eastern Europe and the former Soviet Union since the fall of the</u> Berlin Wall: <u>Review of the Changes in the Environment and Natural Resources</u>
SD	66.2010	Anna Alberini and Milan Ščasný: <u>Context and the VSL: Evidence from a Stated Preference Study in Italy and the Czech Republic</u>
SD	67.2010	Francesco Bosello, Ramiro Parrado and Renato Rosa: <u>The Economic and Environmental Effects of an EU</u> Ban on Illegal Logging Imports. Insights from a CGE Assessment
IM	68.2010	Alessandro Fedele, Paolo M. Panteghini and Sergio Vergalli: <u>Optimal Investment and Financial Strategies</u> under Tax Rate Uncertainty
IM	69.2010	Carlo Cambini, Laura Rondi: <u>Regulatory Independence and Political Interference</u> : <u>Evidence from EU Mixed-</u> <u>Ownership Utilities' Investment and Debt</u>
SD	70.2010	Xavier Pautrel: Environmental Policy, Education and Growth with Finite Lifetime: the Role of Abatement Technology
SD	71.2010	Antoine Leblois and Philippe Quirion: <u>Agricultural Insurances Based on Meteorological Indices:</u> <u>Realizations, Methods and Research Agenda</u>
IM IM	72.2010 73.2010	Bin Dong and Benno Torgler: <u>The Causes of Corruption: Evidence from China</u> Bin Dong and Benno Torgler: <u>The Consequences of Corruption: Evidence from China</u>

IM	74 2010	Fereveloup Verdineiad and Vasaman Gorij: The Oil-Based Economies International Research Project. The
IIVI	74.2010	Case of Iran
CC	75 2010	<u>Case of Irall.</u>
uc	75.2010	Scenes Michaelopoulos, Amitza Ragnavi and Giovanni Prarolo (IXXXII). <u>Trade and Geography III the</u>
SD.	76 2010	Zhone Vienz Zhenzi China in the Transition to a Low Carbon Economy
30	70.2010	Zhong Zhang: <u>Chima in the transition to a Low-Carbon Economy</u>
SD	//.2010	Valentina latolla, Massimiliano Mazzanti and Francesco Nicolli: <u>Are You SURE You Want to Waste Policy</u>
		Chances? Waste Generation, Landfill Diversion and Environmental Policy Effectiveness in the EU15
IM	78.2010	Jean Tirole: <u>Illiquidity and all its Friends</u>
SD	79.2010	Michael Finus and Pedro Pintassilgo: International Environmental Agreements under Uncertainty: Does
		the Veil of Uncertainty Help?
SD	80.2010	Robert W. Hahn and Robert N. Stavins: The Effect of Allowance Allocations on Cap-and-Trade System
		Performance
SD	81 2010	Francisco Alpizar Fredrik Carlsson and Maria Naranio (Ixxxviji): The Effect of Risk Ambiguity and
50	01.2010	Coordination on Farmers' Adaptation to Climate Change: A Framed Field Experiment
۲D	82 2010	Coordination of markets Adaptation to cancer angle. A market field Experiment
30	82.2010	Shardu Agrawaa ahu Maelis Carraro (ixxxviii). Assessing the Role of Micromatice in rostering Adaptation
CD	02 2010	to chimate Change
SD	83.2010	Wolfgang Lutz (IxxxvIII): Improving Education as Key to Enhancing Adaptive Capacity in Developing
		Countries
SD	84.2010	Rasmus Heltberg, Habiba Gitay and Radhika Prabhu (lxxxviii): <u>Community-based Adaptation: Lessons</u>
		from the Development Marketplace 2009 on Adaptation to Climate Change
SD	85.2010	Anna Alberini, Christoph M. Rheinberger, Andrea Leiter, Charles A. McCormick and Andrew Mizrahi:
		What is the Value of Hazardous Weather Forecasts? Evidence from a Survey of Backcountry Skiers
SD	86.2010	Anna Alberini, Milan Ščasný, Dennis Guignet and Stefania Tonin: The Benefits of Contaminated Site
02	00.2010	Cleanup Revisited: The Case of Naples and Caserta Italy
CC	97 2010	David Sorte Mansah William Odwar Erediak Antah Eredua and Stanhan Amisah. Traditional
uc	87.2010	Paul Sario-Mensan, William Oduro, Fredrick Anton Fredua and Stephen Antisan. Traditional
		Representations of the Natural Environment and Biodiversity Conservation: Sacred Groves in Ghana
IM	88.2010	Gian Luca Clementi, Thomas Cooley and Sonia Di Giannatale: <u>A Theory of Firm Decline</u>
IM	89.2010	Gian Luca Clementi and Thomas Cooley: <u>Executive Compensation: Facts</u>
GC	90.2010	Fabio Sabatini: Job Instability and Family Planning: Insights from the Italian Puzzle
SD	91.2010	ZhongXiang Zhang: Copenhagen and Beyond: Reflections on China's Stance and Responses
SD	92.2010	ZhongXiang Zhang: Assessing China's Energy Conservation and Carbon Intensity: How Will the Future
		Differ from the Past?
SD	93,2010	Daron Acemoglu, Philippe Aghion, Leonardo Bursztyn and David Hemous: The Environment and Directed
02	2012010	Technical Change
SD	94 2010	Valaria Costantiai and Massimiliano Mazzanti. On the Green Side of Trade Competitiveness?
30	94.2010	Valeria Costantini and massiminatio mazzanti. On the Green side of trade competitiveness:
15.4	05 2010	Environmental Policies and innovation in the EU
IM	95.2010	Vittoria Cerasi, Barbara Chizzolini and Marc Ivaldi: <u>The Impact of Mergers on the Degree of Competition</u>
		in the Banking Industry
SD	96.2010	Emanuele Massetti and Lea Nicita: The Optimal Climate Policy Portfolio when Knowledge Spills Across
		Sectors
SD	97.2010	Sheila M. Olmstead and Robert N. Stavins: Three Key Elements of Post-2012 International Climate Policy
		Architecture
SD	98.2010	Lawrence H. Goulder and Robert N. Stavins: Interactions between State and Federal Climate Change
		Policies
IM	99 2010	Philippe Aghion John Van Reenen and Luigi Zingales: Innovation and Institutional Ownership
<u> </u>	100 2010	Angele Antoni Eshio Selatini and Auge Sedini. The Selavia Sundrama Seciel Control in a Crawing
uc	100.2010	Angelo Antoci, Fabio Sabatim and Mauro Sodim: <u>The Solaria Syndrome: Social Capital in a Growing</u>
60	101 0010	<u>Hyper-tecnnological Economy</u>
SD	101.2010	Georgios Kossioris, Michael Plexousakis, Anastasios Xepapadeas and Aart de Zeeuw: On the Optimal
		Taxation of Common-Pool Resources
SD	102.2010	ZhongXiang Zhang: Liberalizing Climate-Friendly Goods and Technologies in the WTO: Product Coverage,
		Modalities, Challenges and the Way Forward
SD	103.2010	Gérard Mondello: Risky Activities and Strict Liability Rules: Delegating Safety
GC	104.2010	João Ramos and Benno Torgler: Are Academics Messy? Testing the Broken Windows Theory with a Field
		Experiment in the Work Environment
INA	105 2010	Mauria Cinechai Erangesca Severini Claudio Socci and Posita Pretaroli: The Economic Impact of the
IIVI	103.2010	Construction Made through the Mars Multiplice Assessed
CD	100 2010	Green Certificate Market through the Macro Multiplier Approach
SD	106.2010	Joele Noally: improving the Energy-Efficiency of Buildings: The Impact of Environmental Policy on
		lechnological Innovation
SD	107.2010	Francesca Sanna-Randaccio and Roberta Sestini: <u>The Impact of Unilateral Climate Policy with Endogenous</u>
		Plant Location and Market Size Asymmetry
SD	108.2010	Valeria Costantini, Massimiliano Mozzanti and Anna Montini: Environmental Performance and Regional
		Innovation Spillovers
IM	109.2010	Elena Costantino, Maria Paola Marchello and Cecilia Mezzano: Social Responsibility as a Driver for Local
		Sustainable Development
GC	110 2010	Marco Percoco: Path Dependence Institutions and the Density of Economic Activities: Evidence from
30	110.2010	Italian Citize
SD.	111 0010	Italian Critics
20	111.2010	Sonja S. reelucksingn and Paulo A.L.D. Nunes: <u>Biodiversity valuation in Developing Countries: A Focus</u>
CD	110 0010	on small Island Developing States (SIDS)
50	112.2010	Znongxiang Znang: In What Format and under What Timetrame Would China Take on Climate
		Commitments? A Roadmap to 2050

SD 113.2010 Emanuele Massetti and Fabio Sferra: A Numerical Analysis of Optimal Extraction and Trade of Oil under Climate Policy 114.2010 Nicola Gennaioli, Andrei Shleifer and Robert Vishny: A Numerical Analysis of Optimal Extraction and IM Trade of Oil under Climate Policy GC 115.2010 Romano Piras: Internal Migration Across Italian regions: Macroeconomic Determinants and Accommodating Potential for a Dualistic Economy SD 116.2010 Messan Agbaglah and Lars Ehlers (lxxxix): Overlapping Coalitions, Bargaining and Networks 117.2010 Pascal Billand, Christophe Bravard, Subhadip Chakrabarti and Sudipta Sarangi (Ixxix): Spying in Multi-SD market Oligopolies 118.2010 Roman Chuhay (Ixxxix): Marketing via Friends: Strategic Diffusion of Information in Social Networks with SD Homophily SD 119.2010 Françoise Forges and Ram Orzach (Ixxxix): Core-stable Rings in Second Price Auctions with Common Values SD 120.2010 Markus Kinateder (Ixxxix): The Repeated Prisoner's Dilemma in a Network SD 121.2010 Alexey Kushnir (Ixxxix): Harmful Signaling in Matching Markets 122.2010 Emiliya Lazarova and Dinko Dimitrov (lxxxix): Status-Seeking in Hedonic Games with Heterogeneous SD Players SD 123.2010 Maria Montero (Ixxxix): The Paradox of New Members in the EU Council of Ministers: A Non-cooperative **Bargaining Analysis** SD 124.2010 Leonardo Boncinelli and Paolo Pin (Ixxxix): Stochastic Stability in the Best Shot Game SD 125.2010 Nicolas Quérou (Ixxxix): Group Bargaining and Conflict SD 126.2010 Emily Tanimura (Ixxxix): Diffusion of Innovations on Community Based Small Worlds: the Role of Correlation between Social Spheres 127.2010 Alessandro Tavoni, Maja Schlüter and Simon Levin (Ixxxix): The Survival of the Conformist: Social Pressure SD and Renewable Resource Management SD 128.2010 Norma Olaizola and Federico Valenciano (lxxxix): Information, Stability and Dynamics in Networks under Institutional Constraints 129.2010 Darwin Cortés, Guido Friebel and Darío Maldonado (Ixxxvii): Crime and Education in a Model of GC Information Transmission IM 130.2010 Rosella Levaggi, Michele Moretto and Paolo Pertile: Static and Dynamic Efficiency of Irreversible Health Care Investments under Alternative Payment Rules 131.2010 Robert N. Stavins: The Problem of the Commons: Still Unsettled after 100 Years SD SD 132.2010 Louis-Gaëtan Giraudet and Dominique Finon: On the Road to a Unified Market for Energy Efficiency: The Contribution of White Certificates Schemes SD 133.2010 Melina Barrio and Maria Loureiro: The Impact of Protest Responses in Choice Experiments 134.2010 Vincenzo Denicolò and Christine Halmenschlager: Optimal Patentability Requirements with Fragmented IM **Property Rights** 135.2010 Angelo Antoci, Paolo Russu and Elisa Ticci: Local Communities in front of Big External Investors: An GC **Opportunity or a Risk?** 136.2010 Carlo Carraro and Emanuele Massetti: Beyond Copenhagen: A Realistic Climate Policy in a Fragmented SD World 137.2010 Valentin Przyluski and Stéphane Hallegatte: Climate Change Adaptation, Development, and International SD Financial Support: Lessons from EU Pre-Accession and Solidarity Funds SD 138.2010 Ruslana Rachel Palatnik and Paulo A.L.D. Nunes: Valuation of Linkages between Climate Change, Biodiversity and Productivity of European Agro-Ecosystems 139.2010 Anna Alberini and Milan Ščasný: Does the Cause of Death Matter? The Effect of Dread, Controllability, SD Exposure and Latency on the Vsl

(lxxxvi) This paper was presented at the Conference on "Urban and Regional Economics" organised by the Centre for Economic Policy Research (CEPR) and FEEM, held in Milan on 12-13 October 2009.

(lxxxvii) This paper was presented at the Conference on "Economics of Culture, Institutions and Crime" organised by SUS.DIV, FEEM, University of Padua and CEPR, held in Milan on 20-22 January 2010.

(lxxxviii) This paper was presented at the International Workshop on "The Social Dimension of Adaptation to Climate Change", jointly organized by the International Center for Climate Governance, Centro Euro-Mediterraneo per i Cambiamenti Climatici and Fondazione Eni Enrico Mattei, held in Venice, 18-19 February 2010.

(lxxxix) This paper was presented at the 15th Coalition Theory Network Workshop organised by the Groupement de Recherche en Economie Quantitative d'Aix-Marseille, (GREQAM), held in Marseille, France, on June 17-18, 2010.